Transformer Design and Optimization: A Literature Survey

Eleftherios I. Amoiralis, Member, IEEE, Marina A. Tsili, Member, IEEE, and Antonios G. Kladas, Member, IEEE

Abstract—With the fast-paced changing technologies in the power industry, new references addressing new technologies are coming to the market. Based on this fact, there is an urgent need to keep track of international experiences and activities taking place in the field of modern transformer design. The complexity of transformer design demands reliable and rigorous solution methods. A survey of current research reveals the continued interest in application of advanced techniques for transformer design optimization. This paper conducts a literature survey and reveals general backgrounds of research and developments in the field of transformer design and optimization for the past 35 years, based on more than 420 published articles, 50 transformer books, and 65 standards.

Index Terms—Analytical methods, artificial intelligence, equivalent circuits, experimental methods, hybrid methods, numerical techniques, standards, survey, transformer books, transformer design, transformer design optimization, transformer modeling, transformers.

I. INTRODUCTION

I N the last years, research in the area of transformer design experienced an expansion. Many papers, standards, books, and reports about new models have been published in the technical literature due mostly to the improvement of the computer power availability, new innovative optimization algorithms, and the greater uncertainty level introduced by the power sector deregulation.

Transformer design is a complex task in which engineers have to ensure that compatibility with the imposed specifications is met, while keeping manufacturing costs low. Moreover, the design methodology may vary significantly according to the transformer type (distribution, power or instrument transformer) and its operating frequency (ranging between 50/60 Hz and a few megahertz), while many alterations according to the core constructional characteristics, the cooling method, or the type of the magnetic material may be encountered [1], [2]. This paper provides an overview of research, development, and the application of various computational methods for transformer design, based

Manuscript received November 14, 2008; revised May 10, 2009 Current version published September 23, 2009. This paper is part of the 03ED045 Research Project that is co-financed by E.U.-European Social Fund (75%) and the Greek Ministry of Development-GSRT (25%). Paper no. TPWRD-00839-2008.

E. I. Amoiralis is with the Department of Production Engineering & Management, Technical University of Crete, Chania GR-73100, Greece (e-mail: eamir@tee.gr).

M. A. Tsili and A. G. Kladas are with the Faculty of Electrical & Computer Engineering, National Technical University of Athens, Athens GR-15780, Greece (e-mail: mtsili@central.ntua.gr; kladasel@central.ntua.gr).

Color versions of one or more of the figures in this paper are available online at http://ieeexplore.ieee.org.

Digital Object Identifier 10.1109/TPWRD.2009.2028763

9% Books 10% Journals, Transactions Magazines & 61% PhDs 8% Standards 12% Transactions ■ Conferences Books Journals, Magazines & PhDs Standards

Conferences

Fig. 1. Sources of the literature survey.

on an extensive number of published papers [3]. The referred publications are mainly extracted from IEEE TRANSACTIONS, IEEE Magazines, IEEE Proceedings, IEE/IET (Institution of Electrical Engineers, currently Institution of Engineering and Technology) Proceedings as well as a few, yet very important conferences in an effort to cover the majority of published papers in the transformer design field (Figs. 1 and 2). However, due to the amplitude of this field, such a survey, no matter how comprehensive, cannot be exhaustive. The review is divided into six major sections: 1) Research efforts focusing on the prediction and/or optimization of specific transformer characteristics; 2) techniques adopted for global transformer design optimization, taking into account all of the relevant performance parameters; (3) transformer post-design performance and modeling; 4) standards governing the transformer qualification; 5) recent trends on transformer technology; and 6) transformer books giving to the reader a convenient starting point concerning important aspects of transformer engineering. The references are grouped and presented according to their methodological approach, however, no comparative analysis or specific details of the methodologies are provided in order to keep the survey as compact and comprehensive as possible. The research focuses mainly on power and distribution transformers and other types of transformers operating at low frequencies, and is not expanded to transformers for high frequency applications, apart from several cases of design models and methodologies applicable to this range of frequencies.

The present bibliographical survey will be particularly useful for: 1) transformer designers and researchers engaged in transformer design, optimization, and quality-enhancement activities



2000



🛛 IEEE Trans. Magn. 🔳 IEEE Trans. Power Del. 🖾 Other IEEE Trans.

Fig. 2. Percentage participation of different IEEE Transaction Journals in the overall amount of IEEE Transaction Journals of the survey.

in today's competitive environment; 2) utility engineers who would like to enrich their educational background about the system interaction aspects of transformers in a power system; and 3) undergraduate and postgraduate students who wish to integrate traditional transformer theory with modern computing practices.

The paper is organized as follows: Section II describes the various transformer types that are present in the relevant bibliography as well as the main considerations during the transformer design process. Section III includes the survey overview of research dedicated to transformer characteristics, while Section IV provides an overview of the research conducted on transformer design optimization. Section V lists publications dedicated to modeling transformers for power system studies. Section VI provides a comprehensive overview of transformer standards that are issued by international organizations. Section VII addresses new considerations in transformer design, under the light of recent developments in the electric power industry as well as the technologies involved in transformer construction, providing an up-to-date review of modern trends in transformer design. Section VIII presents a brief description of transformer books. Finally, Section IX concludes this paper.

II. TRANSFORMER DESIGN

A transformer has been defined by ANSI/IEEE [4] as a static electric device consisting of a winding, or two or more coupled windings, with or without a magnetic core, for introducing mutual coupling between electric circuits. Transformers are extensively used in electric power systems to transfer power by electromagnetic induction between circuits at the same frequency, usually with changed values of voltage and current. Transformers are one of the primary components for the transmission and distribution of electrical energy. Their design results mainly from the range of application, the construction, the rated power and the voltage level.

A. Transformer Types

Different kinds of transformer types may be encountered, according to their use, their cooling method or the construction of their magnetic circuit. A major classification is realized according to the power and voltage ratings: transformers with a rated power up to 2.5 MVA and a voltage up to 36 kV are referred to as distribution transformers, while all transformers of higher ratings are classified as power transformers.

In particular, a power transformer has been defined by ANSI/IEEE [4] as a transformer that transfers electric energy in any part of the circuit between the generator and the distribution primary circuits. Power transformers may be further classified according to their scope of application, as described in the following. Transformers that are directly connected to the generator of a power station are called generator transformers. Their power range goes up to far above 1000 MVA and their voltage range extends to approximately 1500 kV. The connection between the different high-voltage system levels is made via network transformers (network interconnecting transformers), which are mainly autotransformers (i.e., transformers where the primary and secondary winding of each phase have a common section). Their power range exceeds 1000 MVA and their voltage range exceeds 1500 kV.

On the other hand-side, distribution transformers are used in the distribution networks in order to transmit energy from the medium voltage network to the low voltage network of the consumers. In particular, a distribution transformer has been defined by ANSI/IEEE [4] as a transformer for transferring electrical energy from a primary distribution circuit to a secondary distribution circuit or consumer's service circuit. In addition, there are various special purpose transformers such as converter transformers, test transformers, instrument transformers or telecommunications transformers, which can be both in the range of power transformers and in the range of distribution transformers as far as rated power and rated voltage are concerned.

As far as the cooling method is concerned, transformers may be designed either as liquid-immersed or dry type transformers. In liquid-immersed transformers, the insulating medium is mineral oil or synthetic insulating liquid while in dry type transformers, the cooling is implemented with natural air circulation. The identification of liquid-immersed transformers according to the cooling method is expressed by a four-letter code (Fig. 3) (IEEE C57.12.00 and IEC 60076-2 standards). The first letter expresses the internal cooling medium in contact with the windings (Letter O designates the mineral oil or synthetic insulating liquid with fire point below 300 °C, letter K designates the insulating liquid with fire point above 300 °C, and letter L designates the insulating liquid with no measurable fire point). The second letter identifies the circulation mechanism for internal cooling medium (Letter N designates the natural convection flow through cooling equipment and in windings, letter F designates the forced circulation through cooling equipment (i.e., coolant pumps) and natural convection flow in windings (also called nondirected flow), and letter D designates the forced circulation through cooling equipment, directed from the cooling equipment into at least the main windings). The third letter expresses the external cooling medium (Letter A designates the air and letter W designates the water). The fourth letter identifies the circulation mechanism for external cooling medium (Letter N designates the natural convection and letter F designates the forced circulation [fans (air cooling) or pumps (water cooling)]). For example, if the internal cooling medium is mineral oil, which is circulated with natural flow, and the external



Fig. 3. Transformer cooling designations.

cooling medium is air, which is circulated with natural convection, then this cooling method is coded as ONAN (Oil Natural Air Natural). In power transformers, various cooling methods are used including oil circulation by pumps, or forced air circulation by fans, or both of the above. As a result, a number of different cooling methods exist: oil natural air forced (ONAF), oil forced air natural (OFAN), oil forced air forced (OFAF), oil forced water forced (OFWF). Combinations such as ONAN/ ONAF, ONAN/OFAN, or ONAN/OFAF are also applicable [5].

Transformer magnetic circuit is constructed in either a shell or a core structure. They are distinguished from each other by the manner in which the primary and secondary coils are placed around the laminated steel core. The shell-type transformer is one where the windings are completely surrounded by transformer steel in the plane of the coil. In core type, the windings surround the laminated iron core. There are two different technologies for stacking the sheets of the magnetic material of the core, providing a further distinction to a) stack-core transformers, where the layers of the sheets of the magnetic material are placed one over the other and the vertical and the horizontal layers are over lapped, and b) wound-core transformers, where the magnetic circuit is of shell type and the sheets are wound. Multi-winding transformers, as well as poly-phase transformers, can be made in either shell or core type designs, with a magnetic circuit that consists of five (in shell-type transformers) or three legs (in core-type transformers), respectively.

B. Transformer Survey Structure

Transformer design must take into account numerous performance parameters and technical constraints. The research in the relevant literature may deal with each one of these parameters separately, or concern the overall transformer optimization. Fig. 4 presents the main categories of the literature survey, which define the structure of the survey overview presented in the next Sections.

III. RESEARCH DEDICATED TO SPECIFIC TRANSFORMER CHARACTERISTICS

The numerous computational methods and engineering models proposed for transformer analysis and the accurate prediction of their characteristics can be roughly categorized into six main groups: 1) numerical techniques (NT) that consist of some of the most widely used tools for transformer simulation. Among the proposed techniques of this group, the finite-element method (FEM) is the most prevalent one; 2) improved versions of the transformer equivalent circuit (EC), in order to include semiempirical descriptions of the core and winding characteristics that affect the accuracy of calculations. The



Fig. 4. Transformer survey structure.

use of the EC is still common in the manufacturing industry, due to its simplicity and its ability to provide reliable results, especially in cases of standardized geometries; 3) analytical methods (AM), employing analytical formulas for the representation of the transformer electromagnetic field as well as other operational characteristics (such as the current distribution), providing alternative modeling with less computational complexity compared to numerical methods; 4) stochastic methods, including artificial-intelligence (AI) techniques, such as genetic algorithms (GAs), which have seen increased usage in the transformer design area over the last few years; 5) experimental methods (EM), combining data provided by measurements with analytical or other methods, in order to provide efficient models for the accurate representation of certain transformer characteristics; 6) hybrid methods (HM) (i.e., combinations of one or more of the methods listed before).

Table I illustrates an overview of the references presented in Sections III-A–III-I, sorted by the subject and methodological approach (based on the six categories defined before). Further details on each reference or group of references are included in the following paragraphs, providing the necessary background for their classification. Since several papers were not entirely dedicated to a single transformer characteristic or methodological approach, their classification was based on the main axes of the proposed methodology or the basic aspects of the addressed research problems. The observation of Table I not only facilitates researchers in the field to categorize previous works but

TABLE I CLASSIFICATION OF REFERENCES PRESENTED IN SECTION III BY SUBJECT AND METHODOLOGICAL APPROACH

	NT		EC		AM
No-load	[7]-[38], [44]	, [45]	[47], [56]-[6	2]	[40], [43]
losses	[83] [101]				[80] [82]
Leakage field	[107]-[123]		[124], [125]		[126]- [129]
Inrush current	[134], [136]		[133], [135],	[137]-	[152], [153]
			[146], [148]· [154]	-[151],	
Dynamic	[155]-[158],	[160],			[163]-[166]
behavior	[161], [167],	[168]			
Noise	[180], [188]		[185]-[187]		[181], [184]
Insulation	[189]-[191]	[223]	[192]-[197]	[247]	[220] [240]
Cooling	[213]-[216],	[223]	[241]-[244],	[247]-	[229]-[240]
DC bias	[271]-[275]		[276]-[280]		
	AI		EM		HM
No-load losses	[48]-[54]	[39], [46]	41], [42], 63]-[79]	[55]	
Load losses		[102]-	[104]		
Leakage field	[130]	[105],	[106], [131],		
		[132]			
Inrush current	[147]	[1(0]		[160]	[1(2]
behavior		[109]		[159]	,[162]
Noise		[170]-	[179], [182],		
		[183]			
Insulation		[199]-	[214]	[198]	
Cooling	[260]- [264]	[265]-	[269]	[219] [228]	-[222], [224]- , [245], [246],
DC bias		[270],	[281]-[283]	[255]	-[259]

also reveals fields that have not yet been covered, proposing further research areas.

A. No-Load Losses

No-load losses are the continuous losses of a transformer, regardless of load, namely, they exist whenever the unit is energized [6]. No-load losses are also called iron or core losses because they are mainly a function of the core materials. The two main components of no-load losses are eddy currents and hysteresis. Hysteresis describes the memory of a magnetic material. More force is necessary to demagnetize magnetic material than it takes to magnetize it; the magnetic domains in the material resist realignment. Eddy current losses are small circulating currents in the core material. The steel core is a conductor that carries an alternating magnetic field, which induces circulating currents in the core. These currents through the resistive conductor generate heat and losses. Cores are typically made from cold-rolled, grain-oriented silicon steel laminations. A third component of core loss is also present, that cannot be directly attributed to eddy current or hysteresis phenomena, and is often called stray, extra or anomalous loss.

FEM has been extensively employed in the no-load losses prediction problem. The application of two-dimensional (2-D) FEM in the calculation of transformer core losses is realized in [7] and [8], where tools for the calculation of stray and eddy losses are developed, while in [9]–[11], hysteresis models are

developed. Detailed modeling of the core magnetic properties based on measurements is presented in [12]–[16], while in [17], eddy current loss in transformers operated by pulsewidth-modulated inverters is presented. Magnetic anisotropy models for use in 2-D numerical analysis are developed in [18]-[20]. Igarashi et al. propose a 2-D FEM method which reduces the number of unknowns in the finite-element analysis of steel laminations in [21]. A numerical method for the calculation of the power losses of nonlinear laminated ferromagnetic cores is developed in [22], performing a two-step analysis (the first step considers the anisotropic conductivity of the material, while the second one introduces its inhomogeneous permeability) and employs FEM models as a tool for the correction and refinement of results provided by the first step. Anisotropic magnetic material properties in conjunction with 2-D FEM are used for the analysis of transformer magnetic material properties frequency dependence [23]. 2-D FEM modeling combined with 2-D and 3-D calculations is carried out in [24] in order to derive a dedicated model of dynamic hysteresis and extra losses in transformer soft magnetic materials. Excessive core losses and temperature rises due to the half-turn effect (the phenomenon where the winding leads are taken out from the different sides of the core leading to an additional half-turn in one of the core windows in a single-phase transformer) are studied through 2-D FEM in [25]. Virtual air gaps created in a transformer core by a change in core permeability in the vicinity of auxiliary windings carrying direct current (dc) are also investigated by 2-D FEM in [26]. In order to achieve more detailed analysis and computation of the transformer magnetic field, the 3-D analysis becomes necessary, as proposed in [27]-[31]. Moreover, Koppikar et al. [32] describe details of statistical analysis used in conjunction with 2-D FEM for quantifying the effect of various factors affecting flitch plate loss along with the results of 3-D FEM simulations carried on slotted and laminated flitch plates. To reduce the computational complexity of 3-D FEM, a practical modeling method of core lamination modeling is investigated in [33]. Nakata et al. investigate the influence of transformer core step lap joints on its loss in [34]. Furthermore, a rigorous analytical study using the finite difference method for magnetic-field calculation is performed by several authors in the literature in order to understand the role of joints in determining the performance of cores of power and distribution transformers [35] and to accurately calculate the 3-D spatial distribution, components, and total core losses in power transformer stacked cores [36], [37]. A dynamic core loss model to estimate core loss in soft ferromagnetic and power ferrite materials with arbitrary flux waveforms, for application in 2-D and 3-D transient finite-element analysis is proposed in [38].

In addition, various studies [39]–[46] explore the local flux distribution in transformer cores as a function of joint design and its relevance for power loss and noise. It is worth noting that although transformer joint air gaps have been well studied using FEM, they are seldom taken into account in circuit models [47].

The application of AI in loss evaluation is addressed in [48] and [49], where the no-load losses as a function of core design parameters are predicted by means of artificial neural networks (ANNs). Georgilakis *et al.* [50], [51] also used ANNs to reduce the iron losses of assembled transformers while optimizing the production process of individual cores using Taguchi

methods. In addition, a combination of three AI methods is presented in [52] and [53], namely, ANNs, decision trees (DTs), and GAs, in order to minimize iron losses during manufacturing of wound-core distribution transformers. In [54], the differential evolution method is applied to determine the magnetically nonlinear characteristics of transformers. Hybrid methods, combining AI techniques with numerical methods, have been employed for the calculation and minimization of core losses, as in [55], where a 3-D permeability tensor FEM is combined with simulated annealing in order to define appropriate design variables of wound cores constructed by a combination of standard and high magnetization grade steel.

Accurate calculation of losses is also provided by the development of improved equivalent circuits, as proposed in [56], where a circuit that is able to predict losses under different operating conditions is presented. Moreover, a final EC of sufficient accuracy, combining hysteresis and eddy current losses contributions, is proposed in [57]. In addition, Adly [58] presented an analytical investigation of the various transformer losses resulting from semirotating flux excitation. The main feature of this analysis is that core magnetic properties are accurately represented and simulated using recently developed vector Preisach-type models of hysteresis. Reference [59] introduces an accurate Preisach-type model of the hysteretic inductor to represent a single-phase transformer for the investigation of the ferroresonance phenomenon, while [60] introduces an accurate transformer core model, using the Preisach theory, to represent the core magnetization characteristic in order to simulate ferroresonance in voltage transformers. An improved magnetic anisotropy model, by use of tensor reluctivity, to accurately express the phase difference between the magnetic flux density and the magnetic-field intensity is developed in [61]. Guerra and Mota present a nonlinear electric circuit to describe the behavior of magnetic cores in low-frequency applications in [62].

Experimental methods, combining data provided by measurements with analytical or other methods, in order to provide efficient models for the accurate representation of hysteresis and power losses in the laminations of power transformers, are proposed in [63] and [64], based on data supplied from the steel manufacturer. Moses [65] reported results in which localized flux density and losses have been measured experimentally in model cores and compared with data obtained from an FEM analysis of the same core geometry. In addition, Girgis et al. [66] carried out an analytical study in an attempt to determine the magnitude of the effects of a number of core production attributes. Albach et al. [67] present a practical method for predicting the core losses in magnetic components for an arbitrary shape of the magnetizing current. Furthermore, Dolinar [68] determined a magnetically nonlinear iron core model of a three-phase three-limb transformer and compared it with the classical saturated iron core model [69]. In [70], an experimental investigation of the factors that influence the harmonic content of magnetizing current (namely the flux density, the degree of saturation and the core stacking technique) is performed.

Many researchers have studied rotational iron losses over the years and a number of techniques for making measurements have been described in the literature. Stranges and Findlay [71] described an apparatus capable of determining iron losses due to rotational flux. In addition, Findlay *et al.* [72] and Davies and

Moses [73] have carried out experiments on various samples in order to test the hypothesis that different stacking patterns of grain oriented silicon steel laminations, cut at small angles to the rolling direction, can reduce the iron core losses in power transformers. Fiorillo et al. have experimentally investigated the magnetic variables of grain oriented alloys, in order to derive models of magnetization curve, hysteresis loops, and losses in any direction [74]. In addition, Marketos and Meydan [75] introduced a novel method of fabricating consolidated stacks of electrical steel into single-phase transformer cores, which can considerably speed up the time required to build the cores by attempting to further reduce the amount of flux that deviates from the rolling direction at the corners of transformer cores. Experimental study of harmonic flux effects in transformer ferromagnetic materials is realized in [76], while a method for predicting the core losses under the sinusoidal conditions with an adequate accuracy from the test results obtained with the real nonsinusoidal voltage waveform is proposed in [77]. Anisotropic behavior of transformer core loss material is experimentally studied and interpreted in [78]. Finally, the silicon steel complex permeability at high frequencies is experimentally determined in [79].

B. Load Losses

Load losses result from load currents flowing through the transformer [6]. Load losses are also called copper or wire or winding losses. The two components of the load losses are the Joule losses (deriving from the product I²R, where symbol I stands for the winding current and symbol R represents the winding resistance) and the stray losses. I²R losses are based on the measured dc resistance, the bulk of which is due to the winding conductors, and the current at a given load. The stray losses are a term given to the accumulation of the additional losses experienced by the transformer, which includes winding eddy losses and losses due to the effects of leakage flux entering internal metallic structures. Auxiliary losses refer to the power required to run auxiliary cooling equipment, such as fans and pumps, and are not typically included in the total losses.

A review of about 50 papers were conducted by Kulkarni and Khaparde [80], which have dealt with one or more components of stray loss from the point of view of estimation and reduction. In this case, stray losses include eddy and circulating current loss in windings, losses in flitch plate, core edge loss, loss due to high current field, and frame and tank losses.

Moreover, a short methodological survey is carried out by Krawczyk and Turowski [81], showing the need of eddy current analysis in electric devices. Robert provides a theoretical discussion about the layer copper factor used in winding loss calculation in [82], focusing on a review of the relevant technical literature.

The FEM analysis is quite commonly used for the eddy loss calculations [83], [84]. Pern and Yeh [85] are engaged in the formulation of a finite-element method based on vector magnetic potential formulation to simulate the electromagnetic field and current distribution in the windings of power transformers with non-negligible circulating current. In addition, the foil-winding eddy loss is estimated by Ram [86], and his model is afterwards exploited for the study of the variation of transformer sheet winding eddy current loss with frequency [87]. Furthermore, the eddy current field due to both windings and heavy current leads in large power transformers is analyzed in [88] by using an improved FEM method based on scalar magnetic potential formulation. This formulation is also employed in [89] for the calculation of three-dimensional distributions of eddy-current densities and loss densities in metals near heavy current leads in a large transformer. On the other hand, vector magnetic potential is used to determine the magnetic field distribution in transformer windings considering inhomogeneous current distribution in [90] and time-periodic nonlinear magnetic fields with eddy currents in [91]. A numerical analysis of losses generated in the tank-wall surrounding the high-current bushings of pad-mounted transformers using a 3-D FEM is conducted in [92]. A special type of finite element is developed in [93] for the calculation of eddy current losses in transformer tank shields. 3-D eddy current calculation is also conducted by means of integral equation method in [94], which, as opposed to 3-D FEM, provides less computational complexity. Analytical representations of the electromagnetic field are also used for the prediction of eddy current losses in transformer tank covers in [95], an investigation that is expanded in [96] by 1) an analytical formulation; 2) a three-dimensional finite-element method; 3) from measured values of the initial temperature rise; and 4) from measured values of the steady-state temperature rise. Eddy losses due to high current leads in transformers are calculated by analytical methods as well as 2-D and 3-D FEM in [97]. In [98], an accurate 3-D formulation to study the boundary eddy current field arising from the heavy winding and terminal lead currents in a compact power transformer is developed. The 3-D reluctance network method is also proposed for transformer load loss prediction in [99] as a fast low cost tool for obtaining power and hottest-spot output data. The same method is employed for the assessment of the effectiveness of a laminated flux collector in controlling the power losses of the transformer in [100]. Finally, FEM has been employed for the investigation of proximity effects on conductor losses in [101].

Experimental study of load losses is also encountered in the relevant literature. When a three-phase three-limb core type transformer is subjected to the load-loss measurement test, it is found that the losses and currents of the phases are asymmetrical (and unequal). Reference [102] attempts to give a comprehensive explanation for the asymmetry in the loss values of the three phases during the load-loss test. Moreover, eddy current losses are experimentally determined in order to analyze the derating of single-phase transformers under nonlinear loads in [103], [104].

C. Leakage Field and Short-Circuit Impedance

The calculation of transformer leakage flux is a prerequisite to the calculation of reactance, short-circuit impedance, shortcircuit forces, and eddy current losses. A review of the most common winding arrangements in dual voltage transformers (i.e., in transformers with primary or/and secondary windings that can be reconnected in order to produce different operating voltages) and their impact on short-circuit strength is carried out in [105]. A further analysis on transformers with tap changers, resulting to variable volts per turn, is carried out in [106], focusing not only to the leakage field but on losses, noise and weight as well.

The finite-element method has been extensively employed in transformer leakage field evaluation. The first research attempts, presented over three decades ago [107], [108], focused on 2-D modeling, due to the restricted performance abilities provided by the early development of personal computers. 2-D FEM remains an efficient tool for leakage field and short-circuit impedance evaluation and more recent developments in the prediction of these parameters are presented in the literature [109]-[111]. Although the 2-D modeling is convenient and useful in some design problems, it can be found insufficient for detailed analysis and computation of the transformer magnetic field, and therefore the 3-D solution, initially proposed by Demerdash et al. at the beginning of 1980 [112], becomes necessary [113]-[117]. In this context, formulas for the exploitation of 3-D FEM model results in winding flux linkage are proposed in [118]. The boundary-element method is another numerical technique extensively used for electromagnetic problems [119]. The main attraction of this method is the simplicity of the data required to solve these problems, along with the high accuracy obtained with boundary elements. Moreover, the combination of boundary and finite elements is another widely used numerical field analysis technique presenting significant advantages in transformer leakage field modeling [120]-[123].

To overcome the main numerical techniques drawback, namely the complexity of the required mesh size, especially in 3-D configurations, alternative leakage field evaluation models have been proposed, with the use of a 3-D reluctance network method [124], [125], falling into the category of EC representation. An alternative method of transformer leakage field calculation is based on simplified analytical formulas, as in [126], where the calculation of self and mutual impedances between sections of transformer windings is performed or [127], where analytical calculations are carried out using Maxwell's differential equations and introducing the vector potential, for the components of the magnetic induction in two-dimensional field space. Analytical methods are often employed by transformer manufacturers in order to simplify the time and complexity of the calculations required in automated design process. Tomczuk and Zakrewski and Tomczuk propose the integral equation method for the calculation of magnetic leakage fields in [128] and [129], respectively.

Stochastic methods are also employed for solving problems of this category, as by Thilagar and Rao [130], who suggested an exact EC model for the estimation of all impedance parameters of three-winding transformers, with the use of GAs. The suggested method also estimates geometrically a complex parameter, that is, mutual leakage between secondary and tertiary windings.

In terms of leakage reactance experimental investigation, the influence of the test circuit (involving low voltage and singlephase excitation) on the respective measurements is presented in [131]. The effects of aluminum magnetic shielding and mild steel magnetic shunts on the leakage flux in the steel tank of a single-phase transformer are experimentally examined in [132].

D. Inrush Current

Transformer inrush currents are high-magnitude, harmonicrich currents generated when transformer cores are driven into saturation during energization. These currents have undesirable effects, including potential damage or loss-of-life to the transformer and reduced power quality on the system [133]. Moreover, a significant impact on transformer winding mechanical stress may be observed [134]. It is also shown that under special conditions, the currents observed after transformer switching on do not contain enough restraining information (e.g., second harmonic), resulting in protective relay misoperation and posing a great problem for protective relaying of power transformers [135]. Inrush current prediction is therefore another important issue during transformer design and various approaches to deal with it are present in the technical literature.

Numerical techniques are present in the above context of inrush current prediction, as in [134] and [136], where 2-D and 3-D FEM is applied to three-legged power transformers for the evaluation of forces on the windings due to inrush current and their comparison to the respective short-circuit forces. However, the majority of the methods used for inrush current simulations are based on the derivation of appropriate equivalent circuits, taking into account the core geometry [137] and structural characteristics [138], the core material nonlinear characteristics [139], [140], using real-time measurements [141], [142], adopting proper parameters for the core magnetic hysteresis Jiles-Atherton model [143] or by proper simulation of the voltage sags caused by inrush currents [144]. Other models take into account the effect of transformer energization to other parallel connected transformers [145]. Frequency domain solution techniques have also been proposed for the simulation of inrush current variation, in order to overcome numerical problems due to the transient nature of the phenomenon [146]. Artificial neural networks have also been employed for the computation of inrush current and the resultant forces on the transformer windings [147].

Practical methods of elimination of transformer inrush current are also proposed in the relevant bibliography as in [148], where proper control strategies of circuit breakers that control transformer switching are presented, or in [149]–[151], by means of sequential phase energization (i.e., by energizing each phase of the transformer in sequence). Other methods may be applied during the design stage, by modifying the winding configuration [152], [153].

Inrush current phenomena may be exploited to estimate transformer performance as in [154] where they are used as a criterion to assess core saturation characteristics.

E. Dynamic Behavior Under Short Circuits and Seismic Stress

The short-circuit current in a transformer creates enormous forces on the turns of the windings. The short-circuit currents in a large transformer are typically 8 to 10 times larger than rated and in a small transformer 20 to 25 times larger than rated. The forces on the windings due to the short-circuit current vary as the square of the current, so whereas the forces at rated current may be only a few newtons, under short-circuit conditions these forces can be tens of thousands of newtons. These mechanical and thermal stresses on the windings must be taken into consideration during the design of the transformer. Transformer behavior under short circuits or internal faults is one of the major concerns during their design, since the ability to overcome the resulting stresses and currents without significant consequences on their operation is a requirement often present in the international technical standards. The accurate representation of this behavior is mainly realized through numerical techniques due to their prevalence in the transformer detailed magnetic field calculation: In [155], 2-D and 3-D FEM are compared in terms of accurate prediction of exerted forces on shell-type transformers under short circuit. In [156], a method was developed to apply 2-D FEM to calculate the parameters for an EC of the transformer with an internal short-circuit fault, while in [157] the method is expanded, representing in detail winding insulation deterioration. Winding internal faults are also modeled by 3-D FEM in [158] and by coupling the finite-element transformer's physical model with external electric circuit equations in [159]. 3-D FEM based on scalar and vector magnetic potential formulation is used for transient eddy current and short-circuit forces estimation, in [160] and [161], respectively.

Equivalent circuit approaches are also employed, although their application is usually coupled to some detailed calculation, often based on numerical methods [162], in order to ensure better representation of the phenomena occurring during short circuits.

Analytical models based on the theory of symmetrical components are also used for the simulation of transformer faults [163], while the stability of continuously transposed cable windings under axial short-circuit forces is investigated through analytical formulas in [164] and the vibration analysis of faulted transformers helical windings is performed in [165]. Analytical methods are proposed in [166] for the calculation of the power flow during transformer internal faults.

Seismic stress is another transient phenomenon with particular interest to transformer engineers, as it can cause severe damage including anchorage failure ripping the transformer case and oil leakage, as well as fracture of porcelain bushings. Therefore, seismic qualification of transformers, focusing on high voltage bushings, is another design consideration that is explored in [167], [168] by means of 3-D FEM analysis, and in [169] by a specially developed experimental setup.

F. Noise

Transformers located near a residential area should have sound level as low as possible. Techniques for power transformer noise control have been proposed over four decades ago [170]–[173] mainly through experimental study and statistical analysis of measurements used to determine the principal factors affecting transformer performance. The sound intensity method for power transformer noise measurements is described and results of its application are demonstrated in [174], while methods based on this technique are also developed in [175] and [176]. In [177] and [178], appropriate conditions for accurate outdoors and indoors measurement of transformer noise are derived and the inherent inaccuracies in the measurements are determined, resulting to the proposition of modifications to the existing IEC and IEEE industry standards of measuring transformer noise.

Since the core magnetic properties and structure are the major factors influencing transformer noise, a lot of research focusses on the analysis and improvement of the core attributes with regard to transformer noise: A simple technique of measuring the dynamic magnetostriction is used to illustrate the effects of improving the stress sensitivity of steel and reducing core vibration by a suitable bonding technique which can also reduce transformer noise in [179]. 3-D FEM structural dynamic analysis is used for the examination of the influence of core lamination upon transformer noise in [180], while vibro-acoustic modelling is also proposed for further analysis. Finally, the relevance of the core magnetic properties for the generation of audible noise in transformer cores is analytically and experimentally investigated in [181].

Methods of noise level reduction are also proposed in the bibliography, through the addition of equipment in transformer substations [182], [183]. More recent approaches perform noise level optimization by means of a reverse calculation method and Linear Programming using an empirical formula for estimating noise levels at the boundary points around the substation premises [184]. The design and the manufacture of a transformer with low sound level require indepth analysis of noise sources. Core, windings and cooling equipment are three main factors of noise, with the first factor the paramount one. To determine a method for the optimum design of the noise-reduction transformer, noise attenuation of a simple structured prototype transformer that utilizes C-cores is quantitatively discussed based on the EC analysis [185], [186]. Similar work is presented in [187]. A recently developed calculation scheme for the computer modelling of the load-controlled noise of liquid-insulated three-phase power transformers is presented in [188]. This modelling scheme allows the precise and efficient computation of the coupled electromagnetic, mechanical and acoustic fields. The equations are solved by using the FEM as well as the boundary-element method (BEM).

G. Insulation

The insulation of a transformer is linked to its ability to withstand surge phenomena and overvoltages likely to occur during its operation. For this purpose, the related work may deal with the analysis of such phenomena, so as to design an adequate transformer insulation system. Other factors that affect transformer insulation life are vibration or mechanical stress, repetitive expansion and contraction, exposure to moisture and other contaminants, and electrical and mechanical stress due to overvoltage and short-circuit currents.

Numerical methods are more scarcely applied for the simulation of the aforementioned phenomena, and are mainly used for the calculation of the transformer electric field [189]–[191]. On the other hand, the majority of research is based on EC representation for transformer analysis under overvoltages, with respect to their geometrical characteristics [192], their dynamic behavior [193], their frequency-response characteristics [194] or the characteristics of the network where they are connected [195]. Other attempts to model the insulation structure of transformers and the quantitative analysis of its dielectric response are also encountered [196], [197]. Hybrid methods, combining finite-element simulations for the derivation of EC parameters have also been proposed [198]. Finally, the exploitation of measurements and the experience from the impact of the operational characteristics on the insulation life assessment is often one of the main methods to design an insulation system and predict its ability to withstand any transient phenomena during the transformer life [199].

Insulation condition assessment is a widely covered topic, and various published works deal with this subject. A theoretical discussion on the aging of oil-impregnated paper in power transformers is carried out in [200], while [201]-[203] and [204] present experimental methods to determine cellulose and oil aging. Transformer oil breakdown is experimentally and theoretically analyzed in [205]. Effects of moisture and aging on the oilpaper insulation of transformers are investigated by return voltage measurements (i.e., the voltage that is built up between the electrodes on a dielectric after the application of direct voltage for a long period of time) in [206]. An analytical model establishing the time to failure of the insulation of transformers given their operating history is developed in [207], based on hourly load and ambient temperature measurements that extract the operating profile of the equipment and IEEE life consumption models to assess the consumed life of insulation. Partial discharge measurements are used to determine dielectric characteristics of transformer oils in [208], [209] and [210]. Frequency response of oil-impregnated pressboard and paper are used for estimating moisture in transformer insulation in [211] and [212]. Experimental investigation of bushing insulation is analyzed in [213]. Methods to overcome the insulating materials degradation with time in service are also proposed, as in [214], where the use of synthetic minerals for the absorption of moisture in paper insulation is discussed.

H. Cooling

Transformer cooling is one of the most important parameters governing a transformer's life expectancy. The total temperature is the sum of the ambient and the temperature rise. The temperature rise in a transformer is intrinsic to that of a transformer at a fixed load. The design of the cooling system is based on the hottest-spot temperature value, and different methods for its prediction are proposed in the literature, along with the overall temperature distribution prediction, according to the transformer cooling method. Furthermore, the improved knowledge of transformer thermal characteristics can allow transformer engineers to achieve enhanced designs and manufacturers are particularly interested in its accurate prediction.

The finite difference method is proposed by Pierce [215] for hottest-spot temperature prediction in dry-type transformers. 2-D FEM thermal calculation is proposed in [216] for the calculation of core hottest-spot temperature in power and distribution transformers. Moreover, it is employed to model the effect of harmonic currents in the winding temperature [217] and to perform heat transfer analysis and obtain the steady state and the transient temperature distribution of SF₆ gas cooled-insulated power transformers [218]. Most recent trends in thermal modeling employ coupled electromagnetic-thermal finite-element models [219], [220]. In [221] a 3-D FEM model using a magnetic scalar potential formulation is combined with a mixed analytical and numerical form of the electrical circuit equation to take into account the skin and proximity effects in the windings, resulting to current densities that are used as inputs to a steady state thermal FEM. In [222] a method that relies on the combination of analytical calculations, 2-D thermal and 3-D electromagnetic FEM is presented for transformer thermal modeling. Rosas *et al.* propose the finite volume method as a means of predicting the improvement of the cooling process of liquid-immersed electrical transformers using heat pipes [223]. For the proper representation of the cooling medium circulation, which affects the transformer thermal performance, thermal FEM is often coupled to computational fluid dynamics (CFD), in 2-D [224], [225] or 3-D models [226], [227]. Coupled fluid flow, heat transfer and electromagnetic numerical analysis is also encountered in the relevant litterature [228], further enhancing the accuracy in the prediction of transformer temperature.

Transformer thermal performance is usually predicted through analytical formulas that use approximations and constants derived from experimental results. Significant research effort is focusing on the improvement of these formulas and the derivation of more efficient calculation methods. In [229] the authors present analytical methods for estimating the temperature and its distribution at different points of the transformer, while in [230] they extend this work, taking into account the thermal inhomogeniety of the windings. This method is later employed to perform an elapsed life assessment study by acquiring insulation-aging data under accelerated thermal stresses in [231]. Calculation methodologies of top oil temperatures and hottest-spot temperatures are evaluated in [232] by comparison to respective measured values in power transformers. Two different thermal analytical models to predict temperature rises in an OFAF transformer at OFAN cooling mode in service are presented and experimentally investigated in [233]. In [234], new thermal loading guides are proposed, based on test results in factories and measured data in the field that are used to determine parameters of thermal differential equations. A mathematical model of the winding hottest-spot response to step changes in the load current of power transformers is presented in [235]. Ryder presents an analytical model to predict winding temperature gradient in power transformers, based on electrical analogy in [236]. The analytical assessment of the impact of ambient temperature rise, as a result of climate change, to distribution transformer loss of life is performed in [237]. In [238] and [239], four top-oil thermal models that require only parameters available from heat-run data and need only measurements utilities routinely monitor are presented and compared. Finally, a short review of developments in analytical thermal models is presented in [240].

Different kinds of equivalent thermal circuits are encountered in the literature, dependent on the geometry of the core and windings and the type of cooling. Appropriate equivalent circuits are developed in [241] for the core rise temperature calculation and power transformer thermal distribution [242]–[244]. In [245] and [246], the parameters of the proposed liquid-immersed transformer thermal model are derived with the use of GAs. Dynamic thermal models are also proposed in the literature [247]–[249], taking proper account of oil viscosity changes and loss variation with temperature, while in [250] a reliability analysis of various dynamic thermal models is carried out. Since transformer cooling is dependent on the hydrodynamic properties of oil, hydraulic models for mass flow distribution have been proposed in the literature, so as to provide detailed representation of the oil flow and pressure in ONAN transformers, as a function of the number and configuration of the cooling ducts [251], [252]. Hydraulic models are combined with heat convection models in [253] and [254].

The variation of transformer loading directly affects its thermal performance and must be correlated to the transformer time constants in order to derive safe conclusions for the resulting thermal loading. Therefore, several works have focused on developing proper load models, suitable for adoption in transformer thermal studies. In [255], a probabilistic model is presented by using load profiles, where variance and covariance are included. A methodology for specifying the winter and summer peak-load limits for substation transformers that carry a temperature-sensitive load, taking into account the random nature of load and ambient temperature as well as their correlation is presented in [256]. Residential loading profiles are extracted in [257], through statistical processing of measured data, proposing a methodology for sizing the transformers to serve these kinds of loads. In [258] a risk-based probabilistic method is presented to assess transformer loading capability, taking into account the probabilistic nature of time-varying loads and ambient temperature. Finally, a method for the evaluation of cyclic loading of power transformers is presented in [259].

Particle swarm method, neural networks, and neurofuzzy networks are also encountered in the relevant research field [260]–[263]. Monte Carlo methods are also used for sensitivity analysis of transformer hottest-spot and equivalent aging in [264].

Experimental investigation of thermal distribution of castresin and liquid-filled transformers is carried out in [265] and [266], respectively. Experimental data are used for the improvement of analytical equations that predict thermal distribution of liquid-filled transformers in [267] and [268]. Transformer oil characteristics before and after modifications of the forced-oil cooling system are experimentally studied in [269].

I. DC Bias

DC current can flow in alternating-current (ac) power lines if a dc potential difference exists between the various grounding points. Such a difference can be caused by a geomagnetic storm [geomagnetically induced current (GIC)] or the injection of dc current by one of the ground electrodes of a dc link [270]. DC flowing through the earthed neutrals of transformer windings causes a dc component in the magnetising current. Owing to nonlinearity, the waveform of this current is strongly distorted. The prediction and impact of this phenomenon has been studied with finite-element method [271]–[275] and equivalent magnetic circuits [276]–[280]. The experimental study of the phenomenon has also been performed, by field tests in a power and distribution transformer in [281] and [282], respectively, and by reactive power measurements in various distribution transformer ratings in [283].

TABLE II CLASSIFICATION OF REFERENCES PRESENTED IN SECTION IV BY PROBLEM TYPE AND PROBLEM NAME

Problem type	Problem name	References
Manufacturing	Transformer design software	[284]-[298]
cost minimization	Transformer design optimization using Deterministic methods	[3], [299]-[305]
	Transformer design optimization using Artificial Intelligence	[3], [306]-[310]
	Transformer design optimization imposing certain technical specifications	[121], [311]- [314]
Operating cost minimization	Energy efficient transformers	[3], [315]-[323]
	Energy efficient transformers taking into account environmental externalities	[3], [324]-[329]

IV. TRANSFORMER DESIGN OPTIMIZATION

Transformer manufacturers use cost optimization techniques during the design phase to minimize material costs and satisfy the utility's loss evaluation requirement. The difficulty in achieving the optimum balance between the transformer cost and performance is a complicated task, and the techniques that are employed for its solution must be able to deal with the design considerations of Section III, so as to provide a design optimum, while remaining cost-effective and flexible. The research associated with design optimization is therefore more restricted involving different mathematical optimization methods.

Whatever the chosen optimization method is, there is also the question of how much detail to include in the problem description. Despite the fact that the goal is to find the lowest cost, one might wish that the solution should provide sufficient information so that an actual design could be produced with little additional work. However, it would be unrealistic to expect that the optimum cost design for a transformer would automatically satisfy all of the mechanical, thermal, and electrical constraints that require sophisticated design algorithms to evaluate. Based on these, our main goal is to present the transformer design optimization techniques that deal with the minimization of the manufacturing as well as operating cost. These techniques are summarized in Table II according to the category of the addressed optimization problem and they are discussed in the forthcoming subsections.

A. Manufacturing Cost Minimization

In optimum design of transformers, the main target is to minimize the manufacturing cost. Therefore, the objective function is a cost function with many terms, including material costs, labor costs, and overhead costs. These component costs, as well as the constraint functions, must be expressed in terms of a basic set of design variables.

In order to compete successfully in a global economy, transformer manufacturers need design software capable of producing manufacturable and optimal designs in a very short time. Traditionally, the transformer design problem has been surrounded by much transformer designer art. The first transformer design was made on computer in 1955 [284]. Later

on, more research in transformer design using computers was pioneered by [285]–[290]. Several design procedures for low-frequency and high-frequency transformers have appeared in the literature after the 70's. Judd and Kressler [291] presented a technique for designing transformers with given size and type of structure to have maximum volt-ampere (VA) output while at the same time insuring the satisfaction of a number of design constraints. The resulting design technique eliminates overdesign problems in that the smallest physical size structure will result consistent with the design objectives. An improved solution of the described problem was presented by Hurley et al. [292]. Poloujadoff et al. [293] show the variation in the price of the transformer depending on the primary turns, which is an approximately hyperbolic function. Also cost curves of the transformer against the magnetic flux density and against the current density are presented. Jeweel [294] does a functional proposal with students in electrical engineering, in which the student designs, builds and tests a 10-VA transformer. Grady et al. [295] deal with the teaching of design of dry type transformers, based on a computer program, where the user optimizes its design based on trial and error. Furthermore, Rubaai [296] describes a computer program yielding an optimal design of a distribution transformer based on user input data (classified in given, independent and dependent). The author includes design transformer formulas used by the program. Andersen [297] presented an optimizing routine, Monica, based on Monte Carlo simulation. Basically, his routine uses random numbers to generate feasible designs from which the lowest cost design is chosen. Hernandez and Arjona [298] develop an object-oriented knowledge-based distribution transformer design system, in conjunction with FEM, which is used as a tool for design performance validation.

Deterministic methods provide robust solutions to the transformer design optimization problem. In this context, the deterministic method of geometric programming has been proposed in [299] in order to deal with the design optimization problem of both low frequency and high frequency transformers. Furthermore, the complex optimum overall transformer design problem, which is formulated as a mixed-integer nonlinear programming problem, by introducing an integrated design optimization methodology based on evolutionary algorithms and numerical electromagnetic and thermal field computations, is addressed in [3], [300]. However, the overall manufacturing cost minimization is scarcely addressed in the technical literature, and the main approaches deal with the cost minimization of specific components such as the magnetic material [301], the no-load loss minimization [302], [303] or the load loss minimization [304]. Techniques that include mathematical models employing analytical formulas, based on design constants and approximations for the calculation of the transformer parameters are often the base of the design process adopted by transformer manufacturers [305].

Apart from deterministic methods, Artificial Intelligence techniques have been extensively used in order to cope with the complex problem of transformer design optimization, such as GAs that have been used for transformer cost minimization [306], performance optimization of cast-resin distribution transformers with stack-core technology [307] or toroidal core transformers [308]. Neural network techniques are also employed as a means of design optimization as in [3], [309] and [310], where they are used for winding material selection and prediction of transformer losses and reactance, respectively.

Another aspect of transformer design optimization consists in providing design solutions in order to maintain certain aspects of transformer performance within the limits imposed by the technical specifications. In this context, the maintenance of short-circuit impedance and losses within the acceptable tolerance is often addressed, as in [121], [311], [312], and [313], [314], respectively.

In a nutshell, it is clear that the transformer design optimization problem remains an active research area.

B. Operating Cost Minimization

Apart from the transformer manufacturing cost, another criterion used for transformer evaluation and optimization is the total owning cost (TOC) taking into account the cost of purchase as well as the cost of energy losses throughout the transformer lifetime [315].

The TOC technique is the most widely used transformer evaluation method for determining the cost-effectiveness of energyefficient transformers, providing a balance between cost of purchase and cost of energy losses. The TOC evaluation method has been developed as a handy tool to reflect the unique financial environment faced by each electric utility when purchasing distribution transformers. According to this method, the variability of the cost of electric energy, capacity and financing costs is expressed through two evaluation factors, called A and B factors, corresponding to the unit cost of no-load and load losses, respectively. It is important to note that the method that defines these two factors varies according to the role of the transformer purchaser in the energy market (two major categories can be considered: electric utilities and industrial users) and the depth of the analysis (depending on the accuracy of the representation of the transformer loading characteristics). It is important to recognize that the perspective of the electric utility is different from the perspective of the industrial and commercial users of transformers. The transformer loss evaluation procedure for the electric utility involves understanding and assessing the total cost of generation, transmission, and distribution transformer losses, while the transformer loss evaluation procedure for an industrial and commercial user requires an understanding and assessment of the electric rates they pay to the electric utility.

An important part of the transformer cost optimization research is devoted to the TOC minimization, as follows. Distribution transformer TOC optimization is analyzed in [316]–[320]. Since the load losses are directly linked to the type of the considered load and the specific details of the network at the transformer installation point, a number of versatile factors should be incorporated in the TOC analysis. Such an analysis is performed in depth in [321], [322].

Furthermore, energy losses of transformers throughout their life cycle increase significantly their operational costs, resulting in TOC values much higher than their purchase price. For the above reason, the decision for what transformer to purchase should not be based only on its purchase price. In general, transformers with the lowest purchase price are also the ones with the highest TOC. Therefore, in order to choose the most economical transformer in the long term, the TOC value during the lifespan of the transformer should be taken into account [323]. Moreover, the external environmental costs [3], [324] should be taken into consideration as well (i.e., the costs that are associated with various types of emissions resulting from the combustion of fossil fuels) so as to compensate for transformer losses.

Recently, the impact of transformer environmental externalities and the contribution of losses to the greenhouse gas emissions generated by the global power generation mix has been addressed [325]–[327]. Furthermore, ways to promote the policy to encourage the use of efficient transformers in the Spanish market are proposed by Frau *et al.* [328], where incentives to private users and electric utilities are introduced, changing Spanish losses regulation, and allowing utilities to participate in the CO_2 emissions market. Moreover, an overview of options available to distribution transformer specifiers, taking advantage of the efficiency and environmental benefits, has been provided [329]. However, a methodology to quantify the impact of environmental externalities on transformer TOC has not yet been developed.

V. POST-DESIGN TRANSFORMER PERFORMANCE

The main incentive of the research presented in the previous sections was to develop models for transformer simulation and adopt methodologies that were able to optimize their performance according to their constructional characteristics, providing several criteria for transformer design optimization. The works presented in this section do not focus on the prediction and evaluation of transformer characteristics during the design stage, but the derivation of proper models for given performance characteristics, to be included in power system studies or other engineering studies, where transformers are involved. The relevant research can be divided into two major categories: models suitable for harmonic load flow studies and models for electromagnetic transient studies. Table III lists the relevant publications by category, further classifying them according to their methodological approach. The acronyms NT, AM, EM, and HM of Table III correspond to the categories defined in Section III (and included in Table I). It must be noted that the majority of the articles concern the derivation of transformer equivalent circuits; therefore, they are all included in the category EC (equivalent circuits) defined in Section III, a classification that is not explicitly denoted in Table III.

A. Harmonic Modeling

The research interest on harmonic load flow studies is continuously growing, due to the increase of nonlinear devices in power systems. Since transformers are key elements in these systems, their modeling is an important subject in harmonic load-flow studies and a number of different approaches have been proposed in the literature. Stensland *et al.* develop a transformer model where the iron and copper losses under low frequency voltage harmonics may be determined either analytically or by FEM, suitable for power system studies [330]. Single and three-phase equivalent circuits taking into account the non-

TABLE III CLASSIFICATION OF REFERENCES PRESENTED IN SECTION V BY SUBJECT AND METHODOLOGICAL APPROACH

	NT	AM	EM	НМ
Harmonic modeling	[330]	[331], [333], [334]		[332]
Transient and dynamic modeling	[354], [355], [377]- [379]	[335], [336], [338]-[353], [356]-[365], [370], [371], [373]-[376]	[369], [372], [380]-[383]	[337], [366]- [368]

linearity of the core are presented in [331]. Mohammed *et al.* implement a transformer transient FEM coupled to external electric circuits and the wavelet packet transfer algorithm for the analysis of harmonic behavior of the transformer currents and the dc load current [332]. In [333], a complete analytical model is developed to calculate the time domain waveform and the harmonic components of the transformer excitation current. Masoum *et al.* develop a nonlinear transformer modeling technique for steady-state operation under unbalanced, asymmetric, and nonsinusoidal operation, capable of computing derating factors in [334].

B. Transient and Dynamic Modeling

One of the weakest components of modern transient simulation software is the transformer model. Many opportunities exist to improve the simulation of its complicated behaviors, which include magnetic saturation of the core, frequency-dependency, capacitive coupling, and topological correctness of core and coil structure [335]. Martinez and Mork present a review of the existing models in [335], providing an overview of their main developments, while Martinez *et al.* provide guidelines for the estimation of transformer model parameters for low- and mid-frequency transient simulations in [336].

In this type of modeling, classical methods to determine transformer magnetic circuit and windings such as the ones described in Section III are combined with frequency and time domain modeling techniques, as presented in the following.

Woivre et al. propose a model for shell-type transformer overvoltage study, where the main R, L, and C parameters are calculated with analytical and numerical methods, while transient overvoltages of the transformer are calculated from frequency response by Fourier transform [337]. Reduction techniques of linear and nonlinear lumped parameters models are proposed by Gutierez et al. [338] and Degeneff et al. [339], respectively. De Leon and Semlyen introduce a simple dynamic hysteresis loss model in [340] and the calculation of transformer EC elementary parameters (namely leakage inductance and capacitance) on a turn-to-turn basis in [341] which are used to derive a winding model suitable for transformer transients in [342]. Moreover, they investigate time domain modeling of eddy currents for electromagnetic transients study in [343] and propose techniques for time step reduction in electromagnetic transient simulation in [344], while in [345], a complete three-phase transformer model is proposed based on their previous works in [341]-[343]. This model is improved in

[346] to include the capacitive effects between turns or sections of a winding. An alternate model for low-frequency electromagnetic transients is proposed in [347] and its more simplified version is presented in [348]. Papadias et al. present three-phase transformer models for the study of switching fast electromagnetic transients in [349]. Distributed equivalent magnetic and electric circuits are introduced in transformer transient analysis in [350]. Tokic et al. develop numerical methods to solve the system of differential equations in state space, describing the transformer transient behavior in [351], while Tokic and Uglesic develop an original method of modeling nonlinear elements, for the elimination of overshooting effects and suppression of numerical oscillations in transformer transient calculations in [352]. Frequency-response analysis is used in [353] to study the transient recovery voltage associated with power transformer terminal faults. Abeywickrama et al. present a 3-D model of electromagnetic (EM)-field distribution in a power transformer at high frequencies for use in frequency-response analysis in [354] and its results are exploited in [355] for high-frequency modeling of power transformers. In [356], the principles of modal analysis are presented, while in [357], modal analysis is used to consider frequency-dependent effects of internal capacitance, inductance, and resistance of windings in order to analyze the transient characteristics of a transformer. In [358], a model is presented for a multiwinding multiphase transformer developed by the nodal inverse inductance matrix, which can be used for transient and steady-state analysis in complicated winding arrangements and network configurations, while in [359] and [360] a multiterminal transformer model is developed for balanced and unbalanced load, respectively. A three-phase transformer dynamic model, providing a good compromise between accuracy and excessive complexity arising in dynamic simulations is presented in [361], while in [362] frequency-dependent time-varying resistance of the transformer winding is considered during modeling the response to lightning impulse wave. A model reference approach for classification of faults that can occur during impulse tests on power transformers is proposed in [363]. Stuehm et al. and Mork develop five-legged wound-core transformer models in [364], [365], while Mork et al. propose a hybrid transformer model based on four typically available sources of information: factory test reports, design data, basic ratings and direct laboratory measurements [366], [367]. In [368], Mork et al. detail the parameter estimation methods developed for the five-legged core of the aforementioned hybrid model. Very fast transient voltage analysis is performed in [369]-[373]. Mombello and Moller present a model with accurate representation of winding losses, developed for the determination of maximal stresses during resonance phenomena within transformers [374], while Mombello performs a deep analysis of the behavior of transformer winding impedances for high frequencies by analyzing the properties of inductance matrices in [375]. In [376], a model that reproduces not only the impedance characteristics seen from each terminal of a core-type distribution transformer but also the surge-transfer characteristics between the primary and secondary sides in a wide range of frequencies is presented.

The coupling of numerical methods with other transient modeling techniques is proposed by other researchers in the field.



ANSI/IEEE Standards III CENELEC Standards IEC Standards

Fig. 5. Percentage participation of IEEE, CENELEC, and IEC standards in the overall amount of standards survey.

An EC based on winding resonances computed by FEM is presented in [377] for the simulation of winding electromagnetic oscillations. Mohammed *et al.* use a circuit coupled FEM analysis under sinusoidal waveforms to derive inductances as a function of the amplitude of ac flux as well as its phase angle during a complete ac cycle for three-phase transformer modeling [378], a work which is later expanded to single-phase transformers [379].

Experimental methods to define models for electromagnetic transient studies are also employed, as in [380], which describes the measurement setup for the extraction of a frequency dependent model of a two winding transformer. A three-phase transformer model including saturation and based on experimental parameters is developed in [381], for the examination of symmetrical and unsymmetrical voltage sag effects on three-legged transformers in [382] and [383], respectively.

VI. TRANSFORMER STANDARDS

A number of transformer relevant standards are listed in this section. These standards are issued by the three institutions: ANSI/IEEE, CENELEC, and IEC (Fig. 5).

IEEE stands for The Institute of Electrical and Electronics Engineers, Inc. IEEE standards are national standards prepared and issued in the United States of America by the IEEE Societies. The use of these standards is wholly voluntary, and it becomes mandatory only when specified in a contractual relationship or when required by a duly constituted legal authority. The IEEE clearly indicates that the existence of an IEEE Standard does not imply that there are no other ways to deal with matters related to the scope of the IEEE standard.

CENELEC stands from Comité Européen de Normalisation Electrotechnique (in English: European Committee for Electrotechnical Standardization). It was created in 1973. CEN-ELEC standards (EN standards) are international standards prepared by working groups and approved by weighted voting among countries being members of the European Union (EU) and other countries included in the European Economic Area Agreement. Furthermore, CENELEC has issued a number of Harmonization Documents (HD), and their formal status is practically the same as that of the EN standards. CENELEC has decided to phase out the Harmonization Documents and replace them by EN standards.

IEC stands for International Electrotechnical Commission, which was officially founded in 1906 in London. The use of IEC standards is a voluntary matter. Technical committee number 14 (TC14) deals with power transformers. The IEC standards do not prescribe how to design and produce transformers. Therefore, it would be meaningless or it would at least be imprecise use of language to say that a transformer shall be or is designed and produced according to IEC standards. The IEC transformer standards establish a series of performance, safety, application, selection and other requirements to be satisfied by the equipment, including performance tests for their certification. In this context, they define certain tests the transformers shall be subjected to before delivery from the factory and state the acceptance criteria. The purpose of the tests is that transformers that have passed these tests shall have good prospects of a long life and high service reliability, when adequately protected and maintained.

It is important to note that between the standards that are issued by these three institutions, there are some basic differences which are not in the scope of this work to emphasize. However, IEC and IEEE have expressed the intention gradually to decrease or remove these basic differences between their standards. It is envisaged that a closer co-operation between these organizations will make the future standardization work more cost efficient.

A. ANSI/IEEE Standards

IEEE Std. C57.12.00-2006—IEEE Standard for Standard General Requirements for Liquid-Immersed Distribution, Power, and Regulating Transformers;

IEEE Std. C57.12.01-2005—IEEE Standard General Requirements for Dry-Type Distribution and Power Transformers, Including Those with Solid-Cast and/or Resin Encapsulated Windings, Revision of C57.12.01-1998;

IEEE Std. C57.12.10-1997—American National Standard for Transformers—230 kV and Below 833/958 through 8333/10 417 kVA, Single-Phase, and 750/862 through 60 000/80 000/100 000 kVA, Three-Phase Without Load Tap Changing; and 3750/4687 through 60 000/80 000/100 000 kVA with Load Tap Changing-Safety Requirement;

IEEE STD C57.12.20-2005—IEEE standard for overheadtype distribution transformers, 500 kVA and smaller: high voltage, 34 500 V and below; low voltage, 7970/13 800y V and below;

IEEE Std. C57.12.21-1992—American National Standard Requirements for Pad-Mounted, Compartmental-Type Self-Cooled, Single-Phase Distribution Transformers with High Voltage Bushings; High-Voltage, 34500 GRYD/19920 Volts and Below; Low-Voltage, 240/120 Volts; 167 kVA and Smaller;

IEEE Std. C57.12.22-1993—American National Standard for Transformers—Pad-Mounted, Compartmental-Type, Self-Cooled Three-Phase Distribution Transformers With High-Voltage Bushings, 2500 kVA and Smaller: High Voltage, 34 500 Grd Y/19 920 Volts and Below; Low Voltage, 480 Volts and Below, Reaffirmed 1998;

IEEE Std. C57.12.23-2002—IEEE Standard for Underground Type, Self-Cooled, Single-Phase, Distribution Transformers with Separable Insulated High-Voltage Connectors; High Voltage 25 000 V and Below; Low Voltage 600 V and Below; 167 kVA and Smaller Revision of C57.12.23-1992;

IEEE Std. C57.12.24-2000 Withdrawn—American National Standard for Transformers Underground-Type Three-Phase Distribution Transformers, 2500 kVA and Smaller; High Voltage, 34 500 GrdY/19 920 Volts and Below; Low Voltage, 480 Volts and Below Requirements; IEEE Std. C57.12.25-1990—American National Standard for Transformers—Pad-mounted, Compartmental-type, Self-cooled, Single-phase Distribution Transformers with Separable Insulated High-voltage Connectors; High Voltage, 34 500 Grdy/19 920 Volts and Below; Low Voltage, 240/120 Volts; 167 kVA and Smaller Requirements;

IEEE Std. C57.12.26-1992—IEEE Standard for Pad-mounted, Compartmental-Type, Self-Cooled, Three-Phase Distribution Transformers for Use with Separable Insulated High-Voltage Connectors (34 500 Grd Y/19 920 V and Below, 2500 kVA and Smaller);

IEEE Std. C57.12.28-2005—IEEE Std. C57.12.28—2005 IEEE Standard for Pad-Mounted Equipment—Enclosure Integrity;

IEEE Std. C57.12.29-2005—IEEE Standard for Pad-Mounted Equipment—Enclosure Integrity for Coastal Environments;

IEEE Std. C57.12.31-2002—IEEE standard for Pole Mounted Equipment—Enclosure Integrity;

IEEE Std. C57.12.32-2002—IEEE Standard for Submersible Equipment—Enclosure Integrity, Reaffirmed 2008;

IEEE Std. C57.12.34-2004—IEEE Standard Requirements for Pad-Mounted, Compartmental-Type, Self-Cooled, three-phase distribution transformers (2500 kVA and smaller)—High-voltage: 34 500 GrdY/19 920 volts and below; low-voltage: 480 volts and below;

IEEE Std. C57.12.35-2007—IEEE Standard for Bar Coding for Distribution Transformers and Step-Voltage Regulators, Revision of C57.12.35-1996;

IEEE Std. C57.12.36-2007—IEEE Standard Requirements for Liquid-Immersed Distribution Substation Transformers;

IEEE Std. C57.12.37-2006—IEEE Standard for the Electronic Reporting of Distribution Transformer Test Data, Revision of 1388-2000;

IEEE Std. C57.12.40-2006—IEEE Standard Requirements for Secondary Network Transformers, Subway and Vault Types (Liquid Immersed);

IEEE Std. C57.12.44-2005—IEEE Standard Requirements for Secondary Network Protectors, Revision of C57.12.44-2000;

IEEE Std. C57.12.50-1981—American National Standard Requirements for Ventilated Dry-Type Distribution Transformers, 1 to 500 kVA, Single-Phase, and 15 to 500 kVA, Three-Phase, with High-Voltage 601 to 34 500 Volts, Low-Voltage 120 to 600 Volts;

IEEE Std. C57.12.51-1981—American National Standard Requirements for Ventilated Dry-Type Power Transformers, 501 kVA and Larger, Three-Phase, with High-Voltage 601 to 34 500 Volts, Low-Voltage 208Y/120 to 4160 Volts;

IEEE Std. C57.12.52-1981—American National Standard Requirements for Sealed Dry-Type Power Transformers, 501 kVA and Larger, Three-Phase, with High-Voltage 601 to 34 500 Volts, Low-Voltage 208Y/120 to 4160 Volts;

IEEE Std. C57.12.55-1987—American National Standard for Transformers—Used in Unit Installations, Including Unit Substations-Conformance Standard;

IEEE Std. C57.12.56-1986—IEEE standard Test Procedure for Thermal Evaluation of Insulation Systems for Ventilated Dry-Type Power and Distribution transformers;

IEEE Std. C57.12.58-1991—IEEE Guide for conducting a transient voltage analysis of a dry-type transformer coil, Reaffirmed 1996, 2002, 2008;

IEEE Std. C57.12.59-2001—IEEE Guide for dr-type transformer through-fault current duration, Reaffirmed 2006;

IEEE Std. C57.12.60-1998—IEEE Guide for test procedures for thermal evaluation of insulation systems for solid-cast and resin-encapsulated power and distribution transformers;

IEEE Std. C57.12.70-2000—IEEE Standard Terminal Markings and Connections for Distribution and Power Transformers;

IEEE Std. C57.12.80-2002—IEEE Standard Terminology for power and distribution transformers, Revision of C57.12.80-1978;

IEEE Std. C57.12.90-2006—IEEE Standard Test Code for Liquid-Immersed Distribution, Power, and Regulating Transformers;

IEEE Std. C57.12.91-2001—IEEE Standard Test Code for Dry-Type Distribution and Power Transformers;

B. CENELEC Standards

EN 60076-1:1997/A12:2002-Part 1: General;

EN 60076-2:1997—Part 2: Temperature rise;

EN 60076-3:2001—Part 3: Insulation levels, dielectric tests and external clearances in air;

EN 60076-4:2002—Part 4: Guide to the lightning impulse and switching impulse testing—Power transformers and reactors;

EN 60076-5:2006—Part 5: Ability to withstand short circuit;

EN 60076-6:2008—Part 6: Reactors;

EN 60076-10:2001—Part 10: Determination of sound levels;

EN 60076-11:2004—Part 11: Dry-type transformers;

EN 60076-13:2006—Part 13: Self-protected liquid-filled transformers;

EN 50216-1:2002—Part 1: General;

EN 50216-2:2002—Part 2: Gas and oil actuated relay for liquid immersed transformers and reactors with conservator;

EN 50216-3:2002—Part 3: Protective relay for hermetically sealed liquid-immersed transformers and reactors without gaseous cushion; EN 50216-4:2002—Part 4: Basic accessories (earthing terminal, drain and filling devices, thermometer pocket, wheel assembly);

EN 50216-5:2002—Part 5: Liquid level, pressure and flow indicators, pressure relief devices and dehydrating breathers;

EN 50216-6:2002—Part 6: Cooling equipment—removable radiators for oil-immersed transformers;

EN 50216-7:2002—Part 7: Electric pumps for transformer oil;

HD 428.1 S1:1992/A1:1995 Three-phase oil-immersed distribution transformers 50 Hz, from 50 to 2500 kVA with highest voltage for equipment not exceeding 36 kV—Part 1: General requirements and requirements for transformers with highest voltage for equipment not exceeding 24 kV;

HD 428.3 S1:1994 Three-phase oil-immersed distribution transformers 50 Hz, from 50 to 2500 kVA, with highest voltage for equipment not exceeding 36 kV—Part 3: Supplementary requirements for transformers with highest voltage for equipment equal to 36 kV;

HD 428.1 S1:1992 Three-phase oil-immersed distribution transformers 50 Hz, from 50 to 2500 kVA with highest voltage for equipment not exceeding 36 kV—Part 1: General requirements and requirements for transformers with highest voltage for equipment not exceeding 24 kV.

C. IEC Standards

IEC 60076-1 (2000-04) Power transformers—Part 1: General;

IEC 60076-2 (1993-04) Power transformers—Part 2: Temperature rise;

IEC 60076-3 (2000-03) Power transformers—Part 3: Insulation levels, dielectric tests and external clearances in air; IEC 60076-4 (2002-06) Power transformers—Part 4: Guide to the lightning impulse and switching impulse testing—power transformers and reactors;

IEC 60076-5 (2006-02) Power transformers—Part 5: Ability to withstand short circuit;

IEC 60076-6 (in process) Power transformers—Part 6: Reactors;

IEC 60076-7 (2005-12) Power transformers—Part 7: Loading guide for oil-immersed power transformers;

IEC 60076-8 (1997-11) Power transformers—Part 8: Application guide;

IEC 60076-10 (2005-07) Power transformers—Part 10: Determination of sound levels;

IEC 60076-10-1 (2005-10) Power transformers—Part 10-1: Determination of sound levels—Application guide; IEC 60076-11 (2004-05) Power transformers—Part 11: Dry-type transformers;

IEC 60076-12 (in process) Power transformers—Part 12: Loading guide for dry-type power transformers;

IEC 60076-13 (2006-05) Power transformers—Part 13: Self-protected liquid-filled transformers;

IEC/TS 60076-14 (2004-11) Power transformers—Part 14: Design and application of liquid-immersed power transformers using high-temperature insulation materials;

IEC 60076-15 (in process) Power transformers—Part 15: Gas-filled power transformers;

TABLE IV CLASSIFICATION OF REFERENCES PRESENTED IN SECTION VII BY PROBLEM TYPE AND PROBLEM NAME

Problem type	Problem name	References
Magnetic circuit	Core losses	[384]-[388]
Windings	Superconducting transformers	[389]-[397]
Insulation	Mineral oil	[398]
	Natural esters or vegetable oils	[399]-[403]
	Accelerated Aging Tests	[404]-[405]
	Mixtures of inslulated liquids	[406]-[408]
	Gas insulated transformers	[409]-[410]
Monitoring and	Transformer fault diagnosis	[411]-[417]
diagnostics	Winding deformation identification	[418]
	Gas analysis	[419]-[421]
	Others	[422]-[425]

VII. RECENT TRENDS IN TRANSFORMER TECHNOLOGY

In the last decade, rapid changes and developments have been made in the field of transformer design. The phenomenal growth of power systems has put tremendous responsibilities on the transformer industry to supply reliable and cost-effective transformers. This section identifies the recent trends in research and development in materials, insulations systems, accessories, and diagnostic techniques, by quoting keyworks that address them, giving pointers to readers desirous of pursuing research in transformers. The references of this Section are summarized in Table IV and they are discussed in further detail in the forthcoming subsections.

A. Magnetic Circuit

There has been a steady development of core steel material in the last century. The trend of reduction in transformer losses in the last few decades is related to a considerable increase in energy costs. One of the ways to reduce the core losses is to use better and thinner grades of core steels, but their price is higher. However, continuous efforts are directed at developing improved electrical steels with lower iron losses for energy-efficient transformers. It is well known that low magnetic losses of amorphous material are attributable to the material's amorphous condition and small thickness of the ribbon [384], [385]. The core losses can be limited by insulating coatings [386], and various types of coatings have been developed for application to both fully processed and semiprocessed electrical steels. In addition, Matsuura et al. [387] and Ezure et al. [388] presented long-term property reliability for iron-based amorphous alloy for use in liquid-immersed transformer cores.

B. Windings

The advent of high-temperature superconducting (HTS) materials has renewed interest in research and development of superconducting transformers. The principal advantages of HTS transformers are: much lower winding material content and losses, higher overload capacity and possibility of coreless design. Some considerations from design point of view are discussed in [389], [390], while in [391] new perspectives of HTS transformer design are introduced. The development of technology based on liquid nitrogen at temperatures up to 79 K

has reduced the complexity and cost of superconducting transformers [392], [393]. A development of a three-phase 100-kVA superconducting transformer with amorphous core has been reported in [394]. A high-superconducting coil that simulated the inner secondary winding of a high-superconducting traction transformer is presented in [395], [396]. Moreover, the magnetization losses in HTS pancake windings according to the operating temperature, is discussed in [397].

C. Insulation

Transformers in electric power distribution and transmission systems are expected to function reliably and efficiently in the long term. The quality of the oil in a transformer plays an important role in performing this function, and the characteristics of transformer oil have been examined and reported on for decades. The majority of transformers use mineral oil in order to meet their cooling demands, due to the fact that mineral oil has not only a low price but also very good electrical insulating properties [398]. However, nowadays, the mineral oil performance cannot meet the modern needs of transformers.

Numerous activities have been initiated to try to improve the properties of mineral oil or to find other substitute liquids Alternatively to conventional mineral oils, natural esters or vegetable oils have been used successfully as transformer dielectric coolants [399]–[403]. Their application offers some advantages, such as safety against a fire incident, environmental friendliness and improved transformer performance.

The results of Accelerated Aging Tests are discussed in length in [404] while in [405] these results suggest that the use of natural esters extends the life of the insulating paper.

Furthermore, mixtures consisting of mineral oil and two other kinds of insulating liquids (namely silicon and synthetic ester oils) are investigated in [406], [407]. Moreover, the "retrofilling" application (i.e., replacing mineral oil with natural esters) is also presented, as a method to extend the remaining thermal life of an aging transformer [408].

Last but not least, there is a considerable progress in the technology of gas immersed transformers in the last decade. Unlike the liquid-immersed transformers, they have SF_6 gas for the insulation and cooling purposes [409], [410].

D. Transformer Monitoring and Diagnostics

Despite the fact that monitoring and diagnostics are not part of the transformer design process, they are relevant to the main design considerations. For this purpose, an overview of some key works dealing with the characteristics presented in the previous Sections is provided in the present Section.

AI techniques and stochastic methods are prevailing in the present category. Neural networks are encountered in the majority of the research dealing with transformer fault diagnosis [411]–[415]. They are also employed for liquid-immersed distribution transformer monitoring in [416]. Furthermore, Stochastic Petri Nets are used for the simulation of the fault diagnosis process of liquid-immersed transformers and the definition of the actions followed to repair the transformer [417]. The particle swarm method is used for winding deformation identification in [418] and dissolved gas analysis in [419]. Finally, fuzzy systems and expert systems are proposed in [420] and [421] for the gas analysis and insulation monitoring, respectively. Significant research effort has focused on discrimination

between fault and inrush current in transformers, and various methods are evolving [422]. Monitoring methods are also based on vibro-acoustic measurements [423], oil temperature [424] or spectral thermography have also been developed. A review of transformer condition assessment methods is provided in [425].

VIII. TRANSFORMER BOOKS

It is necessary to make a brief presentation of transformer books, giving to the reader a convenient starting point. This kind of review will be extremely helpful and handy not only to undergraduate and postgraduate students but also to the transformer industrial engineers. The thorough treatment of all-important aspects of transformer engineering given will provide the reader all the appropriate background to pursue research and development activities in the field of transformers. Bibliography on transformer books is inspired by the IEEE Power Engineering Society (PES) Transformer Committee (http://www.transformerscommittee.org), highlighting a list of 52 books in the domain of transformers.

- James H. Harlow, *Electric Power Transformer Engineering, 2nd Edition*, CRC Press, New York, 2007.
- Giorgio Bertagnolli, Short—Circuit Duty of Power Transformers 3rd Revised Edition, ABB Management Services Ltd. Transformers, Zurich, Switzerland, 2006.
- ABB Transformer and Engineering Services North America, Service Handbook for Power Transformers, ABB, USA, 2006.
- ABB Power Technologies Management Ltd. Transformer Handbook, ABB, Switzerland, 2004.
- ABB Business Area Power Transformers, *Testing of Power Transformers*, ABB, Zurich, 2003.
- Hydroelectric Research and Technical Services Group, *Transformers: Basics, Maintenance, and Diagnostics,* US Department of the Interior, Bureau of Reclamation, Government Printing Office, April 2005.
- Bharat Heavy Electricals Ltd., *Transformers*, Mc-Graw-Hill, New York, 2005.
- S. V. Kulkarni, S. A. Khaparde, *Transformer Engineering Design & Practice*, Marcel Dekker, Inc., New York, 2004.
- M. Horning, J. Kelly, S. Myers, R. Stebbins, *Transformer Maintenance Guide, Third Edition*, Transformer Maintenance Institute, S. D. Myers Inc., 2004.
- Indrajit Dasgupta, *Design of Transformers*, Tata McGraw-Hill Publishing Co. Ltd., New Delhi, 2002.
- John J. Winders Jr., Power Transformers Principles and Applications, Marcel Dekker, Inc., New York, 2002.
- Robert M. Del Vecchio, Bertrand Poulin, Pierre T. Feghali, Dilipkumar M. Shah, Rajendra Ahuja, *Transformer Design Principles—With Application to Core-Form Power Transformers*, Gordon and Breach Science Publishers, Canada, 2001.
- Alexander Publications, editor, *Distribution Transformer Handbook*, Alexander Publications, Newport Beach, California, 2001.
- Axel Krämer, On-Load Tap-Changers for Power Transformers, Operation Principles, Applications and Selection, MR-Publication, Regensburg, Germany, 2000.
- Martin Heathcote, J & P Transformer Book 12th Edition, Butterworth-Heinemann Ltd. Great Britain, 1998.

- Barry W. Kennedy, *Energy Efficient Transformers*, Mc-Graw-Hill Companies, Inc., New York, 1998.
- Alfred Berutti, P.E., Practical Guide to Applying, Installing and Maintaining Transformers, Intertec Publishing Corporation, EC&M Books, Overland Park, Kansas, 1998.
- Stephen L. Herman, Donald E. Singleton, *Delmar's Standard Guide to Transformers*, Delmar Publishers, New York, 1996.
- Eric Lowden, *Practical Transformer Design Handbook* 2nd Edition, Tab Books Inc. Pennsylvania, 1989.
- H. P. Moser, V. Dahinden, *et al. Transformerboard II*, H. AG, Rapperswil, Switzerland, 1987.
- K. Karsai, D. Kerenyi, L. Kiss, Large Power Transformers, (Studies in Electrical and Electronic Engineering, Vol. 25), Elsevier Co., New York, 1987.
- Bernard Hochart, editor, *Power Transformer Handbook*, Butterworths & Co. Ltd., London, 1987.
- A. W. Goldman, C. G. Pebler, Volume 2 *Power Transformers*, Electric Power Research Institute, Palo Alto, California, 1987.
- William M. Flanagan, Handbook of Transformer Design & Applications—2nd Edition, McGraw-Hill Book Co., New York, 1986.
- H. P. Moser, V. Dahinden, *et al.*, *Transformerboard*, H. AG, Rapperswil, Switzerland, 1979.
- R. Feinberg, editor, *Modern Power Transformer Practice*, Halsted Press, 1979.
- Colonel Wm. T. McLyman, *Transformer and Inductor Design Handbook*, Marcel Dekker, Inc., New York, 1978.
- Kenneth L. Gebert, Kenneth R. Edwards, *Transformers Principles and Applications 2nd Edition*, American Technical Publishers, Inc., Illinois, 1974.
- Power Transformer Department, L. F. Blume, A. Boyajian, *Transformer Connections*, General Electric, Schnectady, New York, 1970.
- Petter I. Fergestad, *Transient Oscillations in Transformer* Windings, Naper Boktrykkeri, Kragero, Norway, 1972.
- M. Waters, The Short-Circuit Strength of Power Transformers, Macdonald & Co., London, 1966.
- R. L. Bean, N. Chacken, Jr, H. R. Moore, E. C. Wentz, *Transformers for the Electric Power Industry*, McGraw-Hill Book Co., New York, 1959.
- L. F. Blume, A. Boyajian, G. Camilli, T. C. Lennox, S. Minneci, V. M. Montsinger, *Transformer Engineering—2nd edition*, John Wiley & Sons, Inc., New York, 1951.
- J. B. Gibbs, *Transformer Principles & Practice*, McGraw-Hill Book Co., New York, 1950.
- Eric E. Wild, *Transformers*, Blackie & Son, 2nd Edition, London, 1948.
- Carl H. Dunlap, W. A. Siefert, Frank E. Austin, *Trans*formers Principles and Applications, American Technical Society, Chicago, 1947.
- W. C. Sealey, *Transformers Theory and Construction*, International Textbook Co., Scranton, Pennsylvania, 1946.
- Members of the Staff of the Department of Electrical Engineering Massachusetts Institute of Technology, *Magnetic Circuits and Transformers*, John Wiley & Sons, Inc., New York, 1943.

- J. Rosslyn, *Power Transformers*, Chemical Publishing Co. Inc., New York, 1941.
- L. H. Hill, *Transformers 149C*, International Textbook Co., Scranton, Pennsylvania, 1937.
- David D. Coffin, *Transformers 149B*, International Textbook Co., Scranton, Pennsylvania, 1935
- G. Camilli, *The Testing of Transformers*, General Electric Co., 1929–1930.
- William T. Taylor, *Electricity Supply Transformer Systems and Their Operation*, Charles Griffin and Co. Ltd., London, 1929.
- Emerson G. Reed, *Transformer Construction and Operation*, McGraw-Hill Co., Inc., New York, 1928
- Emerson G. Reed, *Essentials of Transformer Prac*tice—Theory, Design and Operation, D Van Nostrand Co., Inc., New York, 1927.
- Alfred Still, *Principals of Transformer Design*, John Wiley & Sons, Inc., New York, 1919.
- William T. Taylor, *Transformer Practice, Manufacture, Assembling, Connections, Operation and Testing*, McGraw-Hill Book Co., Inc. New York, 1913.
- Frederick Bedell, *The Principals of the Transformer*, The Macmillian Co., New York, 1896.
- Fleming J. A., *The Alternate Current Transformer in Theory and Practice, Volume 1, The Induction of Electric Currents*, New Edition, The Electrician Printing and Publishing Co. Ltd., London, 1896.
- Fleming J. A., *The Alternate Current Transformer in Theory and Practice, Volume II, The Utilization of Induced Currents,* New Edition, The Electrician Printing and Publishing Co. Ltd., London, 1892.
- Caryl D. Haskin, *Transformers*, Bubier Publishing Co., Lynn, Mass., 1892.
- Friedrich Uppenborn, *History of the Transformer*, original translated from German, E & F.N. Spon, New York, 1889. (Reprint available from Kessinger Publishing, Montana.)

IX. CONCLUSION

In this paper, an overview of the literature concerning transformer design has been undertaken, focusing on the progress realized in the past four decades. Relevant publications from international journals have been selected, covering a broad range of engineering methods and design considerations. Moreover, the most important international standards governing transformer performance and qualification requirements have been presented. In addition, a brief presentation of transformer books has been introduced. The difficulties to include and categorize the majority of the research in such a vast field were overcome by a convenient survey structure, taking into account various design considerations. This survey provides important information on the main directions of the considered research and the future trends in the field of transformer design.

REFERENCES

- R. M. Del Vecchio, B. Poulin, P. T. Feghali, D. M. Shah, and R. Ahuja, *Transformer Design Principles-With Application to Core-Form Power Transformers.* New York: Gordon and Breach, 2001.
- [2] S. V. Kulkarni and S. A. Khaparde, *Transformer Engineering Design & Practice*. New York: Marcel Dekker, 2004.

- [3] E. I. Amoiralis, "Energy savings in electric power systems by development of advanced uniform models for the evaluation of transformer manufacturing and operating cost" (in Greek) Ph.D. dissertation, Tech. Univ. Crete, , Greece, 2008. [Online]. Available: http://www.library. tuc.gr/artemis/PD2009-0001/PD2009-0001.pdf
- [4] IEEE Standard Terminology for Power and Distribution Transformers, IEEE Std. C57.12.80, 2002
- [5] Use and Maintenance of ELVIM Oil-Immersed Distribution Transformers. Inofyta, Greece: Schneider Electric AE, 2001.
- [6] J. H. Harlow, Electric Power Transformer Engineering. Boca Raton, FL: CRC, 2004.
- [7] D. Pavlik, D. C. Johnson, and R. S. Girgis, "Calculation and reduction of stray and eddy losses in core-form transformers using a highly accurate finite element modelling technique," IEEE Trans. Power Del., vol. 8, no. 1, pp. 239–244, Jan. 1993.
- [8] S. A. Holland, G. P. O'Connell, and L. Haydock, "Calculating stray losses in power transformers using surface impedance with finite elements," IEEE Trans. Magn., vol. 28, no. 2, pp. 1355-1358, Mar. 1992.
- [9] I. Mayergoyz, "Mathematical models of hysteresis," *IEEE Trans. Magn.*, vol. 22, no. 5, pp. 603–608, Sep. 1986.
- [10] Y. Bernard, E. Mendes, and F. Bouillault, "Dynamic hysteresis modeling based on Preisach model," IEEE Trans. Magn., vol. 38, no. 2, pt. 1, pp. 885–888, Mar. 2002. [11] J. V. Leite, A. Benabou, N. Sadowski, and M. V. F. da Luz, "Finite
- element three-phase transformer modeling taking into account a vector hysteresis model," IEEE Trans. Magn., vol. 45, no. 3, pp. 1716-1719, Mar. 2009.
- [12] M. Enokizono and N. Soda, "Core loss analysis of transformer by improved FEM," J. Mag. Mag. Mat., vol. 196, pp. 910–912, 1999. [13] M. Enokizono, T. Todaka, and K. Nakamura, "Flux distribution in a
- wound core of a single-phase transformer," J. Mag. Mag. Mat., vol. 160, pp. 61-62, 1996.
- [14] C. H. Yu and A. Basak, "Optimum design of transformer cores by analyzing flux and iron loss with the aid of a novel software," *IEEE Trans.* Magn., vol. 29, no. 2, pp. 1446–1449, Mar. 1993.
- [15] A. Basak, C.-H. Yu, and G. Lloyd, "Efficient transformer design by computing core loss using a novel approach," IEEE Trans. Magn., vol. 30, no. 5, pp. 3725–3728, Sep. 1994. [16] M. Enokizono and N. Soda, "Finite element analysis of transformer
- model core with measured reluctivity tensor," IEEE Trans. Magn., vol. 33, no. 5, pp. 4110-4112, Sep. 1997. [17] R. Liu, C. C. Mi, and D. W. Gao, "Modeling of eddy-current loss of
- electrical machines and transformers operated by pulsewidth-modulated inverters," IEEE Trans. Magn., vol. 44, no. 8, pp. 2021-2028, Aug. 2008. [18] H. V. Sande, T. Boonen, I. Podoleanu, F. Henrotte, and K. Hameyer,
- "Simulation of a three-phase transformer using an improved anisotropy model," IEEE Trans. Magn., vol. 40, no. 2, pp. 850-855, Mar. 2004.
- [19] T. Nakata, K. Fujiwara, N. Takahashi, M. Nakano, and N. Okamoto, 'An improved numerical analysis of flux distributions in anisotropic materials," IEEE Trans. Magn., vol. 30, no. 5, pp. 3395-3398, Sep. 1994.
- [20] L. Janicke, A. Kost, R. Merte, T. Nakata, N. Takahashi, K. Fujiwara, and K. Muramatsu, "Numerical modeling for anisotropic magnetic media including saturation effects," IEEE Trans. Magn., vol. 33, no. 2, pp. 1788-1791, Mar. 1997.
- [21] H. Igarashi, K. Watanabe, and A. Kost, "A reduced model for finite element analysis of steel laminations," IEEE Trans. Magn., vol. 42, no. 4, pp. 739–742, Apr. 2006.
- [22] I. Sebestyén, S. Gyimóthy, J. Pávó, and O. Bíró, "Calculation of losses in laminated ferromagnetic materials," IEEE Trans. Magn., vol. 40, no. , pp. 924–927, Mar. 2004.
- [23] S. Ûrata, M. Enokizono, T. Todaka, and H. Shimoji, "The calculation considered two-dimensional vector magnetic properties depending on frequency of transformers," *IEEE Trans. Magn.*, vol. 42, no. 4, pp. 687-690, Apr. 2006.
- [24] O. Maloberti, V. Mazauric, G. Meunier, A. Kedous-Lebouc, and O. Geoffroy, "An energy-based formulation for dynamic hysteresis and extra-losses," *IEEE Trans. Magn.*, vol. 42, no. 4, pp. 895–898, Apr. 2006
- [25] G. B. Kumbhar and S. V. Kulkarni, "Analysis of half-turn effect in power transformers using nonlinear-transient FE formulation," IEEE
- Frans. Power Del., vol. 22, no. 1, pp. 195–200, Jun. 2007.
 [26] A. Konrad and J. F. Brudny, "Virtual air gap length computation with the finite-element method," *IEEE Trans. Magn.*, vol. 43, no. 4, pp. 1829-1832, Apr. 2007.
- [27] N. Demerdash, T. Nehl, O. Mohammed, and F. Fouad, "Nonlinear three dimensional magnetic vector potential finite element solution of field problems including experimental verification," IEEE Trans. Magn., vol. 17, no. 6, pp. 3408-3410, Nov. 1981.
- [28] O. Mohammed, N. Demerdash, and T. Nehl, "Nonlinear three dimensional field computation methods in laminated iron cores under saturated conditions," IEEE Trans. Magn., vol. 19, no. 5, pp. 2091-2093, Sep. 1983.

- [29] R. Wang and N. Demerdash, "A combined vector potential-scalar potential method for FE computation of 3D magnetic fields in electrical devices with iron cores," IEEE Trans. Magn., vol. 27, no. 5, pp. 3971-3977, Sep. 1991.
- [30] C. Lin, C. Xiang, Z. Yanlu, C. Zhingwang, Z. Guoqiang, and Z. Yinhan, "Losses calculation in transformer tie plate using the finite element method," IEEE Trans. Magn., vol. 34, no. 5, pt. 1, pp. 3644-3647, Sep. 1998
- [31] J. V. Leite, A. Benabou, N. Sadowski, S. Clenet, J. P. A. Bastos, and F. Piriou, "Implementation of an anisotropic vector hysteresis model in a 3-D finite-element code," *IEEE Trans. Magn.*, vol. 44, no. 6, pp. 918-921, Jun. 2008.
- [32] D. A. Koppikar, S. V. Kulkarni, P. N. Srinivas, S. A. Khaparde, and R. Jain, "Evaluation of flitch plate losses in power transformers," IEEE Trans. Power Del., vol. 14, no. 3, pp. 996-1001, Jul. 1999.
- [33] S. Nogawa, M. Kuwata, T. Nakau, D. Miyagi, and N. Takahashi, "Study of modeling method of lamination of reactor core," IEEE Trans. Magn., vol. 42, no. 4, pp. 1455–1458, Apr. 2006. [34] T. Nakata, N. Takahashi, and Y. Kawase, "Magnetic performance of
- step lap joints in distribution transformer cores," IEEE Trans. Magn., vol. 18, no. 6, pp. 1055-1057, Nov. 1982.
- [35] G. F. Mechler and R. S. Girgis, "Magnetic flux distributions in transformer core joints," *IEEE Trans. Power Del.*, vol. 15, no. 1, pp. 198-203. Jan. 2000.
- [36] Ed. G. teNyenhuis, G. F. Mechler, and R. S. Girgis, "Flux distribution and core loss calculation for single phase and five limb three phase transformer core designs," IEEE Trans. Power Del., vol. 15, no. 1, pp. 204-209, Jan. 2000.
- [37] G. F. Mechler and R. S. Girgis, "Calculation of spatial loss distribution in stacked power and distribution transformer cores," IEEE Trans. *Power Del.*, vol. 13, no. 2, pp. 532–537, Apr. 1998. [38] D. Lin, P. Zhou, W. N. Fu, Z. Badics, and Z. J. Cendes, "A dynamic
- core loss model for soft ferromagnetic and power ferrite materials in transient finite element analysis," *IEEE Trans. Magn.*, vol. 40, no. 2, pp. 1318-1321, Mar. 2004.
- [39] G. W. Swift, "Excitation current and power loss characteristics for mitered joint power transformer cores," IEEE Trans. Magn., vol. MAG-11, no. 1, pp. 61–64, Jun. 1975. [40] S. Charap and F. Judd, "A core loss model for laminated transformers,"
- IEEE Trans. Magn., vol. 10, no. 3, pp. 678-681, Sep. 1974.
- [41] Z. Valkovic, "Influence of transformer core design on power losses," IEEE Trans. Magn., vol. MAG-18, no. 2, pp. 801-804, Mar. 1982.
- [42] F. Loffler, T. Booth, H. Pfutzner, C. Bengtsson, and K. Gramm, "Relevance of step-lap joints for magnetic characteristics of transformer cores," Proc. Inst. Elect. Eng., Electr. Power Appl., vol. 142, no. 6, pp. 371-378, Nov. 1995.
- [43] Ed. G. teNyenhuis, R. S. Girgis, and G. F. Mechler, "Other factors contributing to the core loss performance of power and distribution transformers," IEEE Trans. Power Del., vol. 16, no. 4, pp. 648-653, Oct. 2001.
- [44] A. Ilo, "Behavior of transformer cores with multistep-lap joints," IEEE Power Eng. Rev., vol. 22, no. 3, pp. 43-47, Mar. 2002.
- [45] A. Mae, K. Harada, Y. Ishihara, and T. Todaka, "A study of characteristic analysis of the three-phase transformer with step-lap wound-core,' IEEE Trans. Magn., vol. 38, no. 2, pp. 829-832, Mar. 2002.
- [46] A. J. Moses and B. Thomas, "Problems in the design of power transformers," IEEE Trans. Magn., vol. MAG-10, no. 2, pp. 148-150, Jun. 1974
- [47] M. Elleuch and M. Poloujadoff, "New transformer model including joint air gaps and lamination anisotropy," IEEE Trans. Magn., vol. 34, no. 5, pp. 3701-3711, Sep. 1998.
- [48] C. Nussbaum, H. Pfützner, T. Booth, N. Baumgartinger, A. Ilo, and M. Clabian, "Neural networks for the prediction of magnetic transformer core characteristics," IEEE Trans. Magn., vol. 36, no. 1, pp. 313-329, Jan. 2000.
- [49] P. S. Georgilakis, N. D. Hatziargyriou, N. D. Doulamis, A. D. Doulamis, and S. D. Kollias, "Prediction of iron losses of wound core distribution transformers based on artificial neural networks," Neurocomput., vol. 23, pp. 15-29, 1998.
- [50] P. Georgilakis, N. Hatziargyriou, D. Paparigas, and S. Elefsiniotis, "Effective use of magnetic materials in transformer manufacturing," J. Mat. Process. Technol., vol. 108, pp. 209–212, 2001. [51] P. S. Georgilakis, N. D. Hatziargyriou, A. D. Doulamis, N. D.
- Doulamis, and S. D. Kollias, "A neural network framework for pre-dicting transformer core losses," in *Proc. 21st IEEE Int. Conf. Power* Industry Computer Applications, May 1999, pp. 301-308.
- [52] P. Georgilakis, N. Hatziargyriou, and D. Paparigas, "AI helps reduce transformer iron losses," IEEE Comput. Appl. Power, vol. 12, no. 4, pp. 41-46, Oct. 1999.
- [53] P. S. Georgilakis, Spotlight on Modern Transformer Design. London, U.K. : Springer, 2009.

- [54] G. Stumberger, S. Seme, B. Stumberger, B. Polajzer, and D. Dolinar, "Determining magnetically nonlinear characteristics of transformers and iron core inductors by differential evolution," IEEE Trans. Magn., vol. 44, no. 6, pp. 1570-1573, Jun. 2008.
- [55] T. D. Kefalas, P. S. Georgilakis, A. G. Kladas, A. T. Souflaris, and D. G. Paparigas, "Multiple grade lamination wound core: A novel technique for transformer iron loss minimization using simulated annealing with restarts and an anisotropy model," IEEE Trans. Magn., vol. 44, no. 6, pp. 1082-1085, Jun. 2008.
- [56] M. Elleuch and M. Poloujadoff, "Analytical model of iron losses in power transformers," IEEE Trans. Magn., vol. 39, no. 2, pp. 973-980, Mar. 2003
- [57] J. Proussalidis, N. Hatziargyriou, and A. Kladas, "Iron lamination efficient representation in power transformers," J. Mat. Process. Technol., vol. 108, pp. 217–220, 2001. [58] A. A. Adly, "Determination of total transformer losses resulting from
- semirotating flux excitation," IEEE Trans. Magn., vol. 31, no. 6, pp. 4253-4255, Nov. 1995.
- [59] A. Rezaei-Zare, M. Sanaye-Pasand, H. Mohseni, S. Farhangi, and R. Iravani, "Analysis of ferroresonance modes in power transformers using Preisach-type hysteretic magnetizing inductance," IEEE Trans. Power Del., vol. 22, no. 2, pp. 919-929, Apr. 2007.
- [60] A. Rezaei-Zare, R. Iravani, M. Sanaye-Pasand, H. Mohseni, and S. Farhangi, "An accurate hysteresis model for ferroresonance analysis of a transformer," IEEE Trans. Power Del., vol. 23, no. 3, pp. 1448-1456, Jul. 2008.
- [61] M. Enokizono, K. Yuki, and S. Kawano, "An improved magnetic field analysis in oriented steel sheet by finite element method considering tensor reluctivity," IEEE Trans. Magn., vol. 31, no. 3, pp. 1797-1800, May 1995.
- [62] F. C. F. Guerra and W. S. Mota, "Magnetic core model," Proc. Inst. Elect. Eng., Sci., Meas. Technol., vol. 1, no. 3, pp. 145-151, May 2007.
- [63] H. Akçay and D. G. Ece, "Modeling of hysteresis and power losses in transformer laminations," *IEEE Trans. Power Del.*, vol. 18, no. 2, pp. 487-492, Apr. 2003.
- [64] R. Arseneau, E. So, and E. Hanique, "Measurements and correction of no-load losses of power transformers," *IEEE Trans. Instrum. Meas.*, vol. 54, no. 2, pp. 503–506, Apr. 2005. [65] A. J. Moses, "Comparison of transformer loss prediction from com-
- puted and measured flux density distribution," IEEE Trans. Magn., vol. 34, no. 4, pp. 1186–1188, Jul. 1998.
- [66] R. S. Girgis, Ed. G. teNijenhuis, K. Gramm, and J.-E. Wrethag, "Experimental investigations on effect of core production attributes on transformer core loss performance," IEEE Trans. Power Del., vol. 13, no. 2, pp. 526-531, Apr. 1998.
- [67] M. Albach, T. Durbaum, and A. Brockmeyer, "Calculating core losses in transformers for arbitrary magnetizing currents: A comparison of different approaches," in Proc. 27th Annu. IEEE Power Electronics Specialists Conf., Jun. 1996, vol. 2, no. 23–27, pp. 1463–1468.
- [68] M. Dolinar, D. Dolinar, G. Stumberger, B. Polajzer, and J. Ritonja, "A three-phase core-type transformer iron core model with included magnetic cross saturation," IEEE Trans. Magn., vol. 42, no. 10, pp. 2849-2851, Oct. 2006.
- [69] J. Pedra, L. Sainz, F. Córcoles, R. Lopez, and M. Salichs, "PSPICE computer model of a nonlinear three-phase three-legged transformer," IEEE Trans. Power Del., vol. 19, no. 1, pp. 200-207, Jun. 2004.
- [70] A. H. Al-Haj and I. El-Amin, "Factors that influence transformer no-load current harmonics," IEEE Trans. Power Del., vol. 15, no. 1, pp. 163–166, Jun. 2000.
- [71] N. Stranges and R. D. Findlay, "Measurement of rotational iron losses in electrical sheet," IEEE Trans. Magn., vol. 36, no. 5, pp. 3457-3459, Sep. 2000.
- [72] R. Findlay, R. Belmans, and D. Mayo, "Influence of the stacking method on the iron losses in power transformers cores," IEEE Trans. Magn., vol. 26, no. 5, pp. 1990-1992, Sep. 1990.
- [73] D. Davies and A. J. Moses, "A novel method of reducing iron losses in power transformer cores," *IEEE Trans. Magn.*, vol. 20, no. 4, pp. 559-562, Jul. 1984.
- [74] F. Fiorillo, L. R. Dupré, C. Appino, and A. M. Rietto, "Comprehensive model of magnetization curve, hysteresis loops, and losses in any direction in grain-oriented Fe-Si," IEEE Trans. Magn., vol. 38, no. 3, pp. 1467-1476, May 2002.
- [75] P. Marketos and T. Meydan, "Novel transformer core design using consolidated stacks of electrical steel," IEEE Trans. Magn., vol. 42, no. 10, pp. 2821-2823, Oct. 2006.
- [76] J. S. Hsu, H. H. Woodson, and S.-S. P. Liou, "Experimental study of harmonic-flux effects in ferromagnetic materials," IEEE Trans. Magn., vol. 25, no. 3, pp. 2678-2685, May 1989.
- [77] T. Tran-Quoc and L. Pierrat, "Correction of the measured core losses under distorted flux," IEEE Trans. Magn., vol. 33, no. 2, pp. 2045-2048, Mar. 1997.

- [78] T. W. Krause, J. A. Szpunar, and D. L. Atherton, "Anisotropic flux density dependence of magnetic Barkhausen noise in oriented 3% Si-Fe steel laminates," IEEE Trans. Magn., vol. 39, no. 1, pp. 562-566, Jan. 2003.
- [79] K. G. N. B. Abeywickrama, T. Daszczynski, Y. V. Serdyuk, and S. M. Gubanski, "Determination of complex permeability of silicon steel for (Bubaliski, Determination of complex permeability of sine see 1701 use in high-frequency modeling of power transformers," *IEEE Trans. Magn.*, vol. 44, no. 4, pp. 438–444, Apr. 2008.
 [80] S. V. Kulkarni and S. A. Khaparde, "Stray loss evaluation in power transformers—A review," in *Proc. IEEE Power Eng. Soc. Winter transformers*, 2009.
- Meeting, Jan. 2000, vol. 3, pp. 2269-2274.
- [81] A. Krawczyk and J. Turowski, "Recent development in eddy current analysis," IEEE Trans. Magn., vol. 23, no. 5, pp. 3032-3037, Sep. 1987.
- F. Robert, "A theoretical discussion about the layer copper factor used in winding losses calculation," IEEE Trans. Magn., vol. 38, no. 5, pp. 3177–3179, Sep. 2002. [83] S. Wang, S.-C. Ji, and Y. Li, "The study of eddy current losses in
- coaxially insulated windings of power transformer," in Proc. Int. Conf. Power System Technology, Oct. 2002, vol. 3, pp. 1392–1395.
- [84] D. A. Koppikar, S. V. Kulkarni, G. Ghosh, S. M. Ainapure, and J. S. Bhavsar, "Circulating-current loss in transformer windings," Proc. Inst. Elect. Eng., Sci. Meas. Tech., vol. 145, no. 4, pp. 136-140, Jul. 1998.
- [85] J.-F. Pern and S.-N. Yeh, "Calculating the current distribution in power transformer windings using finite element analysis with circuit constraints," Proc. Inst. Elect. Eng., Sci. Meas. Technol., vol. 142, no. 3, pp. 231–236, May 1995.
- [86] B. S. Ram, "Loss and current distribution in foil windings of transformers," Proc. Inst. Elect. Eng., Gen. Transm. Distr., vol. 145, no. 6, pp. 709-716, Nov. 1998.
- [87] B. S. Ram, "Variation of transformer sheet winding eddy current loss with frequency," IEEE Power Eng. Rev., vol. 21, no. 10, pp. 49-50, Oct. 2001.
- [88] T. Renyuan, L. Yan, L. Feng, and T. Lijian, "Resultant magnetic fields due to both windings and heavy current leads in large power transformers," *IEEE Trans. Magn.*, vol. 32, no. 3, pp. 1641–1644, May 1996.
- R. Tang, "Computation of 3-D open boundary eddy-current fields in large transformers using T-W-Fm method coupled with GTM," *IEEE* [89] Trans. Magn., vol. 38, no. 2, pp. 1257-1260, Mar. 2002
- [90] K. Iwata and H. Hirano, "Magnetic field analysis by edge elements of magnetic components considering inhomogeneous current distribution within conductor windings," IEEE Trans. Magn., vol. 42, no. 5, pp. 1549-1554, May 2006.
- [91] T. Nakata, N. Takahashi, K. Fujiwara, and A. Ahagon, "3-D non-linear eddy current analysis using the time-periodic finite element method,' *IEEE Trans. Magn.*, vol. 25, no. 5, pp. 4150–4152, Sep. 1989.
 J. C. Olivares, R. Escarela, S. V. Kulkarni, F. de León, E. Melgoza-
- Vasquez, and O. Hernández-Anaya, "Improved insert geometry for reducing tank-wall losses in pad-mounted transformers," IEEE Trans. Power Del., vol. 19, no. 3, pp. 1120–1126, Jul. 2005.
 [93] C. Guerin, G. Tanneau, G. Meunier, and P. Labie, "A shell element
- for computing 3D eddy currents—Application to transformers," *IEEE Trans. Magn.*, vol. 31, no. 3, pp. 1360–1363, May 1995.
 [94] Y. Higuchi and M. Koizumi, "Integral equation method with surface
- impedance model for 3D eddy current analysis in transformers," IEEE *Trans. Magn.*, vol. 36, no. 4, pp. 774–779, Jul. 2000. [95] J. Turowski and A. Pelikant, "Eddy current losses and hot-spot eval-
- uation in cover plates of power transformers," Proc. Inst. Elect. Eng., *Electr. Power Appl.*, vol. 144, no. 6, pp. 435–440, Nov. 1997. [96] S. V. Kulkarni, J. C. Olivares, R. Escarela-Perez, V. K. Lakhiani, and
- J. Turowski, "Evaluation of eddy current losses in the cover plates of distribution transformers," *Proc. Inst. Elect. Eng., Sci. Meas. Technol.*, vol. 151, no. 5, pp. 313–318, Sep. 2004.
 [97] D. A. Koppikar, S. V. Kulkarni, S. A. Khaparde, and S. K. Jha, "Eval-
- uation of eddy losses due to high current leads in transformers," Proc. Inst. Elect. Eng., Sci. Meas. Technol., vol. 144, no. 1, pp. 34-38, Jan.
- [98] S. L. Ho, Y. Li, R. Y. Tang, K. W. E. Cheng, and S. Y. Yang, "Calcula-[96] S. E. Ho, T. Li, K. T. Tang, K. W. E. Cheng, and S. T. Tang, Calculation of eddy current field in the ascending flange for the bushings and tank wall of a large power transformer," *IEEE Trans. Magn.*, vol. 44, no. 6, pp. 1522–1525, Jun. 2008.
 [99] D. A. Koppikar, S. V. Kulkarni, and J. Turowski, "Fast 3-dimensional field and hence increase methics for the field and hence increase methics for the set of the set of the set of the set."
- interactive computation of stray field and losses in asymmetric transformers," Proc. Inst. Elect. Eng., Gen. Transm. Distrib., vol. 147, no. 4, pp. 197-201, Jul. 2000.
- [100] M. Rizzo, A. Savini, and J. Turowski, "Influence of flux collectors on stray losses in transformers," IEEE Trans. Magn., vol. 36, no. 4, pp. 1915-1918, Jul. 2000.
- [101] M. Vitelli, "Numerical evaluation of 2-D proximity effect conductor losses," IEEE Trans. Power Del., vol. 19, no. 3, pp. 1291-1298, Jul. 2004.
- [102] R. Escarela-Perez, S. V. Kulkarni, N. K. Kodela, and J. C. Olivares-Galvan, "Asymmetry during load-loss measurement of three-phase three-limb transformers," IEEE Trans. Power Del., vol. 22, no. 3, pp. 1566-1574, Jul. 2007.

- [103] E. F. Fuchs, D. Yildirim, and W. M. Grady, "Measurement of eddycurrent loss coefficient PEC-R, derating of single-phase transformers, and comparison with K-factor approach," *IEEE Trans. Power Del.*, vol. 15, no. 1, pp. 148-154, Jun. 2000.
- [104] D. Yildirim and E. F. Fuchs, "Measured transformer derating and com-parison with harmonic loss factor (FHL) approach," *IEEE Trans. Power Del.*, vol. 15, no. 1, pp. 186–191, Jun. 2000. [105] J. A. Ebert, "Criteria for reliable dual voltage power transformers,"
- IEEE Trans. Power Del., vol. 10, no. 2, pp. 845-852, Apr. 1995.
- [106] J. A. Ebert, "Application considerations for variable volts per turn power transformers," IEEE Trans. Power Del., vol. 9, no. 1, pp. . 240–248, Jan. 1994.
- [107] O. W. Andersen, "Transformer leakage flux program based on the finite element method," *IEEE Trans. Power App. Sys.*, vol. PAS-92, no. 2, pp. 682-689, Mar. 1973
- [108] O. W. Andersen, "PC-based field calculations for electric power applications," IEEE Comput. Appl. Power, vol. 2, no. 4, pp. 22-25, Oct. 1989
- [109] X. Cui and G. Zhang, "Automatic design of impedance in shell-form power transformer," *IEEE Trans. Magn.*, vol. 36, no. 4, pp. 1822–1825, Jul. 2000.
- [110] A. Lindblom, J. Isberg, and H. Bernhoff, "Calculating the coupling factor in a multilayer coaxial transformer with air core," IEEE Trans.
- *Magn.*, vol. 40, no. 5, pp. 3244–3248, Sep. 2004.
 [111] L. Xiaosong, W. Suping, Z. Yusheng, H. Gui, and L. Guzong, "Influence of high temperature superconducting transformer geometry on leakage magnetic field," *IEEE Trans. Magn.*, vol. 44, no. 4, pp. 1024024 492-496, Apr. 2008.
- [112] N. Demerdash, T. Nehl, and F. Fouad, "Finite element formulation and analysis of three dimensional magnetic field problems," IEEE Trans.
- [113] A. G. Kladas, M. P. Papadopoulos, and J. A. Tegopoulos, "Leakage flux and force calculation on power transformer windings under shortcircuit: 2D and 3D models based on the theory of images and the finite element method compared to measurements," IEEE Trans. Magn., vol. 30, no. 5, pt. 2, pp. 3487–3490, Sep. 1994. [114] L. Xiaosong, W. Suping, Z. Yusheng, H. Gui, and L. Guzong, "In-
- fluence of high temperature superconducting transformer geometry on leakage magnetic field," *IEEE Trans. Magn.*, vol. 44, no. 4, pp. 492-496, Apr. 2008.
- [115] M. Djurovic and C. J. Carpenter, "3-dimensional computation of transformer leakage fields and associated losses," IEEE Trans. Magn., vol. no. 5, pp. 1535–1537, Sep. 1975.
 K. Zakrewski and M. Kukaniszyn, "Three-dimensional model of one-
- [116] R. Zattevisht and R. Ratinsz J., "Inter dimensional model of one and three-phase transformer for leakage field calculation," *IEEE Trans. Magn.*, vol. 28, no. 2, pp. 1344–1347, Mar. 1992.
 [117] M. A. Tsili, A. G. Kladas, P. S. Georgilakis, A. T. Souflaris, and D. G. Paparigas, "Advanced design methodology for single and dual voltage
- wound core power transformers based on a particular finite element model," Electric Power Syst. Res., vol. 76, pp. 729-741, 2006.
- [118] O. Biro, "Computation of the flux linkage of windings from magnetic scalar potential finite element solutions," *Proc.Inst. Elect. Eng., Sci. Meas. Tech.*, vol. 149, no. 5, pp. 182–185, Sep. 2002.
- [119] C. A. Brebbia and R. Magureanu, "The boundary element method for electromagnetic problems," *Eng. Anal.*, vol. 4, no. 4, pp. 178–185, 1987
- [120] G. Meunier, J. L. Coulomb, S. J. Salon, and L. Krahenbul, "Hybrid finite element boundary element solutions for three dimensional scalar potential problems," IEEE Trans. Magn., vol. 22, no. 5, pp. 1040-1042, Sep. 1986.
- [121] M. A. Tsili, "Development of mixed finite element-boundary element numerical techniques for the design of power transformers" Ph.D. Thesis (in Greek), Nat. Tech. Univ. Athens, Athens, Greece, 2005 [Online]. Available: http://artemis.cslab.ntua.gr/Dienst/UI/1.0/Download/artemis.ntua.ece/PD2005-0005
- [122] M. A. Tsili, A. G. Kladas, P. S. Georgilakis, A. T. Souflaris, C. P. Pitsilis, J. A. Bakopoulos, and D. G. Paparigas, "Hybrid numerical techniques for power transformer modeling: A comparative analysis validated by measurements," IEEE Trans. Magn., vol. 40, no. 2, pp. 842-845, Mar. 2004.
- [123] M. Tsili, A. Kladas, P. Georgilakis, A. Souflaris, and D. Paparigas, Numerical techniques for design and modeling of distribution transformers," J. Mat. Process. Tech., vol. 161, pp. 320–326, 2005. [124] J. Turowski, M. Turowski, and M. Kopec, "Method of three-dimen-
- sional network solution of leakage field of three-phase transformers," *IEEE Trans. Magn.*, vol. 26, no. 5, pp. 2911–2919, Sep. 1990. [125] M. Elleuch and M. Poloujadoff, "A contribution to the modeling of
- three phase transformers using reluctances," *IEEE Trans. Magn.*, vol. 32, no. 2, pp. 335–343, Mar. 1996.
- [126] D. J. Wilcox, W. G. Hurley, and M. Conlon, "Calculation of self and mutual impedances between sections of transformer windings," Proc. Inst. Elect. Eng., Gen. Transm. Distrib., vol. 136, no. 5, pp. 308-314, Sep. 1989.

- [127] P. Raitsios, "Leakage field of a transformer under conventional and superconducting condition," J. Mat. Process. Tech., vol. 108, pp. 246-252, 2001
- [128] B. Tomczuk, "Analysis of 3-D magnetic fields in high leakage reactance transformers," IEEE Trans. Magn., vol. 30, no. 5, pp. 2734-2738, Sep. 1994.
- [129] K. Zakrewski and B. Tomczuk, "Magnetic field analysis and leakage inductance calculation in current transformers by means of 3-D integral methods," IEEE Trans. Magn., vol. 32, no. 3, pp. 1637-1640, May 1996
- [130] S. H. Thilagar and G. S. Rao, "Parameter estimation of three-winding transformers using genetic algorithm," Eng. Appl. Artificial Intell., vol. 15, no. 5, pp. 429-437, Sep. 2002.
- [131] M. F. Lachman and Y. N. Shafir, "Influence of single-phase excitation and magnetizing reactance on transformer leakage reactance measurement," IEEE Trans. Power Del., vol. 12, no. 4, pp. 1538-1546, Oct. 1997
- [132] A. Basak and H. G. Kendall, "Leakage flux in the steel tank of a 2.5-kVA single phase transformer," IEEE Trans. Magn., vol. 23, no. 5, pp. 3831-3835, Sep. 1987.
- [133] J. H. Brunke and K. J. Fröhlich, "Elimination of transformer inrush currents by controlled switching-Part I: Theoretical considerations," IEEE Trans. Power Del., vol. 16, no. 2, pp. 276-280, Apr. 2001
- [134] M. Steurer and K. Fröhlich, "The impact of inrush currents on the mechanical stress of high voltage power transformer coils," *IEEE Trans. Power Del.*, vol. 17, no. 1, pp. 155–160, Jan. 2002.
- [135] A. Wiszniewski, W. Rebizant, D. Bejmert, and L. Schiel, "Ultrasaturation phenomenon in power transformers-Myths and reality," IEEE Trans. Power Del., vol. 23, no. 3, pp. 1327-1334, Jul. 2008.
- J. Faiz, B. M. Ebrahimi, and T. Noori, "Three- and two-dimensional [136] finite-element computation of inrush current and short-circuit electromagnetic forces on windings of a three-phase core-type power transformer," *IEEE Trans. Magn.*, vol. 44, no. 5, pp. 590–597, May 2008. [137] L. Sáinz, F. Córcoles, J. Pedra, and L. Guasch, "Theoretical calculation
- of inrush currents in three- and five-legged core transformers," IEEE Trans. Power Del., vol. 22, no. 2, pp. 986-995, Apr. 2007.
- [138] S.-D. Chen, R.-L. Lin, and C.-K. Cheng, "Magnetizing inrush model of transformers based on structure parameters," IEEE Trans. Power Del., vol. 20, no. 3, pp. 1947-1954, Jul. 2005.
- S. G. Abdulsalam, W. Xu, and V. Dinavahi, "Modelling and simula-[139] tion of three-phase transformers for inrush current studies," Proc. Inst. Elect. Eng., Gen. Transm. Distrib., vol. 152, no. 3, pp. 328-333, May 2005.
- [140] M. G. Vanti, S. L. Bertoli, S. H. L. Cabral, A. G. Gerent, and P. P. Kuo, "Semianalytic solution for a simple model of inrush currents in transformers," IEEE Trans. Magn., vol. 44, no. 6, pp. 1270-1273, Jun. 2008
- [141] C. E. Lin, C. L. Cheng, C. L. Huang, and J. C. Yeh, "Investigation of magnetizing inrush current in transformers. Part I-Numerical simulation," *IEEE Trans. Power Del.*, vol. 8, no. 1, pp. 246–254, Jan. 1993. [142] C. E. Lin, C. L. Cheng, C. L. Huang, and J. C. Yeh, "Investigation of
- magnetizing inrush current in transformers. Part II-Harmonic analysis," IEEE Trans. Power Del., vol. 8, no. 1, pp. 255-263, Jan. 1993.
- [143] X. Wang, D. W. P. Thomas, M. Sumner, J. Paul, and S. H. L. Cabral, "Characteristics of Jiles-Atherton model parameters and their application to transformer inrush current simulation," IEEE Trans. Magn., vol. 44, no. 3, pp. 340–345, Mar. 2008. [144] M. Nagpal, T. G. Martinich, A. Moshref, K. Morison, and P. Kundur,
- "Assessing and limiting impact of transformer inrush current on power quality," IEEE Trans. Power Del., vol. 21, no. 2, pp. 890-896, Apr. 2006.
- [145] H. Bronzeado and R. Yacamini, "Phenomenon of sympathetic interaction between transformers caused by inrush transients," Proc. Inst.
- [146] J. J. Rico, E. Acha, and M. Madrigal, "The study of inrush current phenomenon using operational matrices," *IEEE Trans. Power Del.*, vol. 16, no. 2, pp. 231–237, Apr. 2001.
 [147] A. A. Adly, "Computation of inrush current forces on transformer windings," *IEEE Trans. Magn.*, vol. 37, no. 4, pp. 2855–2857, Lul.
- windings," IEEE Trans. Magn., vol. 37, no. 4, pp. 2855-2857, Jul. 2001
- [148] J. H. Brunke and K. J. Fröhlich, "Elimination of transformer inrush currents by controlled switching-Part II: Application and performance considerations," IEEE Trans. Power Del., vol. 16, no. 2, pp. 281-285, Apr. 2001.
- [149] Y. Cui, S. G. Abdulsalam, S. Chen, and W. Xu, "A sequential phase energization technique for transformer inrush current reduction—Part I: Simulation and experimental results," *IEEE Trans. Power Del.*, vol. 20, no. 2, pt. 1, pp. 943-949, Apr. 2005.
- W. Xu, S. G. Abdulsalam, Y. Cui, and X. Liu, "A sequential phase [150] energization technique for transformer inrush current reduction—Part II: Theoretical analysis and design guide," *IEEE Trans. Power Del.*, vol. 20, no. 2, pt. 1, pp. 950-957, Apr. 2005.

- [151] S. G. Abdulsalam and W. Xu, "A sequential phase energization method for transformer inrush current reduction—Transient performance and practical considerations," *IEEE Trans. Power Del.*, vol. 22, no. 1, pp. 208–216, Jun. 2007.
- [152] J. F. Chen, T. J. Liang, C. K. Cheng, S. D. Chen, R. L. Lin, and W. H. Yang, "Asymmetrical winding configuration to reduce inrush current with appropriate short-circuit current in transformer," *Proc. Inst. Elect. Eng., Electr. Power Appl.*, vol. 152, no. 3, pp. 605–611, May 2005.
 [153] C. K. Cheng, T. J. Liang, J. F. Chen, S. D. Chen, and W. H. Yang,
- [153] C. K. Cheng, T. J. Liang, J. F. Chen, S. D. Chen, and W. H. Yang, "Novel approach to reducing the inrush current of a power transformer," *Proc. Inst. Elect. Eng., Electr. Power Appl.*, vol. 151, no. 3, pp. 289–295, May 2004.
- [154] S. G. Abdulsalam, W. Xu, W. L. A. Neves, and X. Liu, "Estimation of transformer saturation characteristics from inrush current waveforms," *IEEE Trans. Power Del.*, vol. 21, no. 1, pp. 170–177, Jan. 2006.
- [155] S. Salon, B. LaMattina, and K. Sivasubramaniam, "Comparison of assumptions in computation of short circuit forces in transformers," *IEEE Trans. Magn.*, vol. 36, no. 5, pp. 3521–3523, Sep. 2000.
 [156] H. Wang and K. L. Butler, "Finite element analysis of internal winding
- [156] H. Wang and K. L. Butler, "Finite element analysis of internal winding faults in distribution transformers," *IEEE Trans. Power Del.*, vol. 16, no. 3, pp. 422–427, Jul. 2001.
- [157] H. Wang and K. L. Butler, "Modeling transformers with internal incipient faults," *IEEE Trans. Power Del.*, vol. 17, no. 2, pp. 500–509, Apr. 2002.
- [158] K. Zakrzewski, B. Tomczuk, and D. Koteras, "Simulation of forces and 3-D field arising during power autotransformer fault due to electric arc in HV winding," *IEEE Trans. Magn.*, vol. 38, no. 2, pp. 1153–1156, Mar. 2002.
- [159] N. Y. Abed and O. A. Mohammed, "Modeling and characterization of transformers internal faults using finite element and discrete wavelet transforms," *IEEE Trans. Magn.*, vol. 43, no. 4, pp. 1425–1428, Apr. 2007.
- [160] T. Renyuan, L. Yan, L. Dake, and T. Lijian, "Numerical calculation of 3D transient eddy current field and short circuit electromagnetic force in large transformers," *IEEE Trans. Magn.*, vol. 28, no. 2, pp. 1418–1421, Mar. 1992.
- 1418–1421, Mar. 1992.
 [161] S. L. Ho, Y. Li, H. C. Wong, S. H. Wang, and R. Y. Tang, "Numerical simulation of transient force and eddy current loss in a 720-MVA power transformer," *IEEE Trans. Magn.*, vol. 40, no. 2, pp. 687–690, Mar. 2004.
- [162] G. B. Kumbhar and S. V. Kulkarni, "Analysis of short-circuit performance of split-winding transformer using coupled field-circuit approach," *IEEE Trans. Power Del.*, vol. 22, no. 2, pp. 936–943, Apr. 2007.
- [163] G. Díaz, P. Arboleya, and J. Gómez-Aleixandre, "Analytical approach to internal fault simulation in power transformers based on fault-related incremental currents," *IEEE Trans. Power Del.*, vol. 21, no. 1, pp. 142–149, Jan. 2006.
- [164] M. R. Patel, "Instability of the continuously transposed cable under axial short-circuit forces in transformers," *IEEE Trans. Power Del.*, vol. 17, no. 1, pp. 149–154, Jan. 2002.
- [165] P. S. Shin, J. L. Jung, and W. Ha, "A free vibration analysis of helical windings of power transformer by pseudospectral method," *IEEE Trans. Magn.*, vol. 43, no. 6, pp. 2657–2659, Jun. 2007.
- [166] M. Jablonski and E. Napieralska-Juszczak, "Internal faults in power transformers," *Proc. Inst. Elect. Eng., Electr. Power Appl.*, vol. 1, no. 1, pp. 105–111, Jan. 2007.
- [167] S. Bellorini, F. Bettinali, M. Salvetti, and G. Zafferani, "Seismic qualification of transformer high voltage bushings," *IEEE Trans. Power Del.*, vol. 13, no. 4, pp. 1208–1213, Oct. 1998.
 [168] S. Ersoy and M. A. Saadeghvaziri, "Seismic response of transformer-
- [168] S. Ersoy and M. A. Saadeghvaziri, "Seismic response of transformerbushing systems," *IEEE Trans. Power Del.*, vol. 19, no. 1, pp. 131–137, Jan. 2004.
- [169] N. Murota, M. Q. Feng, and G. Y. Liu, "Earthquake simulator testing of base-isolated power transformers," *IEEE Trans. Power Del.*, vol. 21, no. 3, pp. 1291–1299, Jul. 2006.
- [170] C. Gordon, "Techniques for the control of transformer noise (A)," J. Acoust. Soc. Am., vol. 36, no. 5, pp. 1037–1037, May 1964.
- [171] C. Gordon, "A method for predicting the audible noise emissions from large outdoors power transformers," *IEEE Trans. Power App. Sys.*, vol. PAS-98, no. 3, pp. 1109–1112, May 1979.
- [172] I. Ver, D. Andersen, M. Myles, and A. Teplitzky, "Field study of sound radiation by power transformers," *IEEE Trans. Power App. Sys.*, vol. PAS-100, no. 7, pp. 3513–3524, Jul. 1981.
- [173] R. Hemmes and D. Graham, "Measurement of self cooled transformer sound levels in relatively high ambients," *IEEE Trans. Power App. Sys.*, vol. PAS-89, no. 7, pp. 1657–1662, Sep. 1970.
 [174] R. Kendig and S. Wright, "Validation of acoustic intensity measure-
- [174] R. Kendig and S. Wright, "Validation of acoustic intensity measurements for power transformers," *IEEE Trans. Power Del.*, vol. 6, no. 4, pp. 1524–1530, Oct. 1991.
- [175] G. Usry, P. Saha, J. Hadden, and A. Pierce, "Prediction of far field sound radiation from transformers," *IEEE Trans. Power App. Sys.*, vol. PAS-99, no. 1, pp. 358–364, Jan. 1980.

- [176] Y. Champoux, B. Gosselin, and J. Nicolas, "Application of the intensity technique to the characterization of transformer noise," *IEEE Trans. Power Del.*, vol. 3, no. 4, pp. 1802–1808, Oct. 1988.
- [177] R. Girgis, K. Garner, M. Bernesjo, and J. Anger, "Measuring no-load and load noise of power transformers using the sound pressure and sound intensity methods—Part-I: Outdoors measurements," in *Proc. IEEE Power Energy Soc. General Meeting—Conversion and Delivery of Electrical Energy in the 21st Century*, Jul. 2008, pp. 1–8.
 [178] R. Girgis, K. Garner, M. Bernesjo, J. Anger, and D. Chu, "Measuring
- [178] R. Girgis, K. Garner, M. Bernesjo, J. Anger, and D. Chu, "Measuring no-load and load noise of power transformers using the sound pressure and sound intensity methods—Part-II: Indoors measurements," in *Proc. IEEE Power and Energy Soc. General Meeting–Conversion and Delivery of Electrical Energy in the 21st Century*, Jul. 2008, pp. 1–8.
- [179] A. Moses, "Measurement of magnetostriction and vibration with regard to transformer noise," *IEEE Trans. Magn.*, vol. 10, no. 2, pp. 154–156, Jun. 1974.
- [180] R. S. Masti, W. Desmet, and W. Heylen, "On the influence of core laminations upon power transformer noise," in *Proc. Int. Conf. Noise* & Vibration Engineering, Sep. 2004, pp. 3851–3862.
- [181] B. Weiser, H. Pfützner, and J. Anger, "Relevance of magnetostriction and forces for the generation of audible noise of transformer cores," *IEEE Trans. Magn.*, vol. 36, no. 5, pp. 3759–3777, Sep. 2000.
- [182] P. Adkins, "Transformer noise reduction with a close-fitting mass-law enclosure," *IEEE Trans. Power App. Syst.*, vol. 82, no. 66, pp. 302–308, Jun. 1963.
- [183] M. Kanoi, Y. Hori, M. Maejima, and T. Obata, "Transformer noise reduction with new sound insulation panel," *IEEE Trans. Power App. Syst.*, vol. PAS-102, no. 9, pp. 2817–2825, Sep. 1983.
- [184] H. Tanaka, D. Yamashita, T. Niimura, and R. Yokoyama, "Transformer noise level optimization based on reverse calculation problem and linear programming," in *Proc. IEEE Power Energy Soc. General Meeting–Conversion and Delivery of Electrical Energy in the 21st Century*, Jul. 2008, pp. 1–7.
- [185] T. Yanada, S. Minowa, O. Ichinokura, and S. Kikuchi, "Design and analysis of noise-reduction transformer based on equivalent circuit," *IEEE Trans. Magn.*, vol. 34, no. 4, pp. 1351–1353, Jul. 1998.
- [186] W. W. L. Keerthipala, Z. RuJing, T. Eu, L. Chionh, and C. Jinn, "Electronic circuits for active control of acoustic noise generated by high voltage transformers," in *Proc. Power Electronic Drives and Energy Systems for Industrial Growth*, 1998, vol. 1, pp. 243–248.
- [187] C. Teoh, K. Soh, R. Zhou, D. Tien, and V. Chan, "Active noise control of transformer noise," in *Proc. Int. Conf. Energy Management and Power Delivery*, Mar. 1998, vol. 2, pp. 747–753.
- [188] M. Rausch, M. Kaltenbacher, H. Landes, R. Lerch, J. Anger, and J. Gerth, "Combination of finite and boundary element methods in investigation and prediction of load-controlled noise of power transformers," *J. Sound Vibr.*, vol. 250, no. 2, pp. 323–338, 2002.
- [189] E. Lesniewska, "The use of 3-D electric field analysis and the analytical approach for improvement of a combined instrument transformer insulation system," *IEEE Trans. Magn.*, vol. 38, no. 2, pp. 1233–1236, Mar. 2002.
- [190] J. Faiz and M. Ojaghi, "Novel techniques for treating singularity problems in the boundary element method applied to electric field evaluation within the tank of a power transformer," *IEEE Trans. Power Del.*, vol. 15, no. 2, pp. 592–598, Apr. 2000.
- [191] A. Khaligh and M. Vakilian, "Power transformers internal insulation design improvements using electric field analysis through finite-element methods," *IEEE Trans. Magn.*, vol. 44, no. 2, pp. 273–278, Feb. 2008.
- [192] D. D. Mairs, D. L. Stuehm, and B. A. Mork, "Overvoltages on fivelegged core transformers on rural electric systems," *IEEE Trans. Ind. Appl.*, vol. 25, no. 2, pp. 366–370, Mar. 1989.
- [193] M. Hori, M. Nishioka, Y. Ikeda, K. Noguchi, K. Kajimura, H. Motoyama, and T. Kawamura, "Internal winding failure due to resonance overvoltage in distribution transformer caused by winter lightning," *IEEE Trans. Power Del.*, vol. 21, no. 3, pp. 1600–1606, Jul. 2006.
- [194] M. Hori, M. Mitani, H. Sakurano, H. Motoyama, and T. Kawamura, "Observation and analysis of incident surge voltage waveforms in substations by winter lightning," *IEEE Trans. Power Del.*, vol. 22, no. 1, pp. 316–322, Jan. 2007.
- [195] M. Popov and L. van der Sluis, "Improved calculations for no-load transformer switching surges," *IEEE Trans. Power Del.*, vol. 16, no. 3, pp. 401–408, Jul. 2001.
- [196] T. K. Saha, P. Purkait, and F. Müller, "Deriving an equivalent circuit of transformers insulation for understanding the dielectric response measurements," *IEEE Trans. Power Del.*, vol. 20, no. 1, pp. 149–157, Jan. 2005.
- [197] T. K. Saha, "Review of time-domain polarization measurements for assessing insulation condition in aged transformers," *IEEE Trans. Power Del.*, vol. 18, no. 4, pp. 1293–1301, Oct. 2003.

- [198] H. Yamashita, V. Cingoski, E. Nakamae, A. Namera, and H. Kitamura, "Design improvements on graded insulation of power transformers using transient electric field analysis and visualization technique," IEEE Trans. Energy Convers., vol. 14, no. 4, pp. 1379-1384, Dec. 1999
- [199] P. M. Balma, R. C. Degeneff, H. R. Moore, and L. B. Wagenaar, "The effects of long term operation and system conditions on the dielectric capability and insulation coordination of large power transformers," IEEE Trans. Power Del., vol. 14, no. 3, pp. 960-971, Jul. 1999.
- [200] L. E. Lundgaard, W. Hansen, D. Linhjell, and T. J. Painter, "Aging of oil-impregnated paper in power transformers," IEEE Trans. Power Del., vol. 19, no. 1, pp. 230-239, Jun. 2004.
- [201] A. J. Kachler and I. Höhlein, "Aging of cellulose at transformer service temperatures. Part 1: Influence of type of oil and air on the degree of polymerization of pressboard, dissolved gases, and furanic compounds in oil," *IEEE Electr. Insul. Mag.*, vol. 21, no. 2, pp. 15–21, Mar. 2005. [202] I. Höhlein and A. J. Kachler, "Aging of cellulose at transformer ser-
- vice temperatures. Part 2. Influence of moisture and temperature on degree of polymerization and formation of furanic compounds in freebreathing systems," IEEE Electr. Insul. Mag., vol. 21, no. 5, pp. 20-24, Sep. 2005.
- [203] T. K. Saha, M. Darveniza, D. J. T. Hill, G. Young, and Z. T. Yao, "Investigating the effects of oxidation and thermal degradation on electrical and chemical properties of power transformers insulation," IEEE Trans. Power Del., vol. 14, no. 4, pp. 1359-1367, Oct. 1999.
- [204] R. M. Morais, W. A. Mannheimer, M. Carballeira, and J. C. Noualhaguet, "Furfural analysis for assessing degradation of thermally up-graded papers in transformer insulation," *IEEE Trans. Dielectr. Electr. Insul.*, vol. 6, no. 2, pp. 159–163, Apr. 1999. [205] R. M. Del Vecchio, "Geometric effects in the electrical breakdown of
- transformer oil," IEEE Trans. Power Del., vol. 19, no. 2, pp. 652-656, Apr. 2004.
- [206] T. Kumar, S. Zheng, and T. Yao, "Experience with return voltage measurements for assessing insulation conditions in service-aged transformers," IEEE Trans. Power Del., vol. 18, no. 1, pp. 128-135, Jan. 2003
- [207] K. T. Muthanna, A. Sarkar, K. Das, and K. Waldner, "Transformer insulation life assessment," IEEE Trans. Power Del., vol. 21, no. 1, pp. 150-156, Jan. 2006.
- [208] M. Pompili, C. Mazzetti, and R. Bartnikas, "PD pulse burst characteristics of transformer oils," IEEE Trans. Power Del., vol. 21, no. 2, pp.
- 689–698, Apr. 2006. [209] M. Pompili, C. Mazzetti, and R. Bartnikas, "Partial discharge pulse sequence patterns and cavity development times in transformer oils under ac conditions," *IEEE Trans. Dielect. Electr. Insul.*, vol. 12, no. 2, pp. 395-403, Apr. 2005.
- [210] X. Wang, B. Li, Z. Xiao, O. L. Russo, H. T. Roman, K. Chin, and K. R. Farmer, "Acoustic energy shifting in transformer oil at different temperatures," *IEEE Trans. Power Del.*, vol. 20, no. 3, pp. 2356–2357, Jul. 2005.
- [211] C. Ekanayake, S. M. Gubanski, A. Graczkowski, and K. Walczak, "Frequency response of oil impregnated pressboard and paper samples for estimating moisture in transformer insulation," IEEE Trans. Power *Del.*, vol. 21, no. 3, pp. 1309–1317, Jul. 2006. [212] K. G. Nilanga, B. Abeywickrama, Y. V. Serdyuk, and S. M. Gubanski,
- 'Exploring possibilities for characterization of power transformer insulation by frequency response analysis (FRA)," IEEE Trans. Power Del., vol. 21, no. 3, pp. 1375–1382, Jul. 2006. [213] S. Zhang, "Analysis of some measurement issues in bushing power
- factor tests in the field," IEEE Trans. Power Del., vol. 21, no. 3, pp. 1350-1356, Jul. 2006.
- [214] I. Fofana, V. Wasserberg, H. Borsi, E. Gockenbach, and M. Farzaneh, 'Drying of transformer insulation using zeolite," IEEE Electr. Insul. Mag., vol. 20, no. 1, pp. 20-30, Jan./Feb. 2004.
- [215] L. Pierce, "Predicting hottest spot temperatures in ventilated dry type transformer windings," *IEEE Trans. Power Del.*, vol. 9, no. 2, pp. 1160-1172, Apr. 1994.
- [216] Ed. G. teNyenhuis, R. S. Girgis, G. F. Mechler, and G. Zhou, "Calculation of core hot-spot temperature in power and distribution trans-formers," *IEEE Trans. Power Del.*, vol. 17, no. 4, pp. 991–995, Oct. 2002
- [217] M. D. Hwang, W. M. Grady, and H. W. Sanders, "Calculation of winding temperatures in distribution transformers subjected to harmonic currents," IEEE Trans. Power Del., vol. 3, no. 3, pp. 1074-1079, Jul. 1988.
- [218] M. B. Eteiba, M. M. A. Aziz, and J. H. Shazly, "Heat conduction prob-lems in SF6 gas cooled-insulated power transformers solved by the finite-element method," IEEE Trans. Power Del., vol. 23, no. 3, pp. 1457-1463, July. 2008.
- [219] K. Preis, O. Bíró, G. Buchgraber, and I. Ticar, "Thermal-electromagnetic coupling in the finite-element simulation of power transformers, IEEE Trans. Magn., vol. 42, no. 4, pp. 999-1002, Apr. 2006.

- [220] C. C. Hwang, P. H. Tang, and Y. H. Jiang, "Thermal analysis of highfrequency transformers using finite elements coupled with temperature rise method," Proc. Inst. Elect. Eng., Electr. Power Appl., vol. 152, no. 4, pp. 832-836, Jul. 2005.
- [221] A. Lefèvre, L. Miègeville, J. Fouladgar, and G. Olivier, "3-D computation of transformers overheating under nonlinear loads," IEEE Trans. Magn., vol. 41, no. 5, pp. 1564–1567, May 2005. [222] M. A. Tsili, E. I. Amoiralis, A. G. Kladas, and A. T. Souflaris, "Hybrid
- numerical-analytical technique for power transformer thermal modeling," IEEE Trans. Magn., vol. 45, no. 3, pp. 1408–1411, Mar. 2009.
- [223] C. Rosas, N. Moraga, V. Bubnovich, and R. Fischer, "Improvement of pipes," *IEEE Trans. Power Del.*, vol. 20, no. 3, pp. 1955–1961, Jul. 2005.
- [224] N. El Wakil, N.-C. Chereches, and J. Padet, "Numerical study of heat transfer and fluid flow in a power transformer," Int. J. Therm. Sci., vol. 45, pp. 615-626, 2006.
- [225] J. M. Mufuta and E. van den Bulck, "Modelling of the mixed convection in the windings of a disc-type power transformer," Appl. Therm. Eng., vol. 20, pp. 417-437, 2000.
- [226] J. Smolka, A. J. Nowak, and L. C. Wrobel, "Numerical modelling of thermal processes in an electrical transformer dipped into polymerised resin by using commercial CFD package fluent," *Comput. Fluids*, vol. 33, pp. 859-868, 2004.
- [227] J. Smolka, D. B. Ingham, L. Elliott, and A. J. Nowak, "Enhanced numerical model of performance of an encapsulated three-phase transformer in laboratory environment," Appl. Therm. Eng., vol. 27, pp. 156-166, 2007.
- [228] J. Smolka and A. J. Nowak, "Experimental validation of the coupled fluid flow, heat transfer and electromagnetic numerical model of the medium-power dry-type electrical transformer," Int. J. Therm. Sci., vol. 47, pp. 1393-1410, 2008.
- [229] M. K. Pradhan and T. S. Ramu, "Prediction of hottest spot temperature (HST) in power and station transformers," *IEEE Trans. Power Del.*, vol. 18, no. 4, pp. 1275–1283, Oct. 2003.
- [230] M. K. Pradhan and T. S. Ramu, "Estimation of the hottest spot temperature (HST) in power transformers considering thermal inhomoge-niety of the windings," *IEEE Trans. Power Del.*, vol. 19, no. 4, pp. 1704-1712, Oct. 2004.
- [231] M. K. Pradhan and T. S. Ramu, "On the estimation of elapsed life of oil-immersed power transformers," *IEEE Trans. Power Del.*, vol. 20, no. 3, pp. 1962–1969, Jul. 2005.
- [232] J. A. Jardini, J. L. P. Brittes, L. C. Magrini, M. A. Bini, and J. Yasuoka, "Power transformer temperature evaluation for overloading conditions," IEEE Trans. Power Del., vol. 20, no. 1, pp. 179-184, Jan. 2005.
- [233] D. Susa, J. Palola, M. Lehtonen, and M. Hyvärinen, "Temperature rises in an OFAF transformer at OFAN cooling mode in service," IEEE Trans. Power Del., vol. 20, no. 4, pp. 2517-2525, Oct. 2005.
- [234] S. Tojo, "Proposition of individual loading guide for power trans-formers," *IEEE Trans. Power Del.*, vol. 21, no. 3, pp. 1383–1389, Jul. 2006.
- [235] H. Nordman, N. Rafsback, and D. Susa, "Temperature responses to step changes in the load current of power transformers," IEEE Trans. Power Del., vol. 18, no. 4, pp. 1110-1117, Oct. 2003.
- [236] S. A. Ryder, "A simple method for calculating winding temperature gradient in power transformers," IEEE Trans. Power Del., vol. 17, no. 4, pp. 977-982, Oct. 2002.
- [237] J. W. Stahlhut, G. T. Heydt, and N. J. Selover, "A preliminary assessment of the impact of ambient temperature rise on distribution transformer loss of life," IEEE Trans. Power Del., vol. 23, no. 4, pp. 2000-2007, Oct. 2008
- 2000–2007, Oct. 2008.
 [238] L. Jauregui-Rivera and D. J. Tylavsky, "Acceptability of four transformer top-oil thermal models—Part I: Defining metrics," *IEEE Trans. Power Del.*, vol. 23, no. 2, pp. 860–865, Apr. 2008.
 [239] L. Jauregui-Rivera and D. J. Tylavsky, "Acceptability of four transformer top-oil thermal models—Part II: Comparing metrics," *IEEE Trans. Power Del.*, vol. 23, no. 2, pp. 866–872, Apr. 2008.
 [240] M. F. Lachman, P. J. Griffin, W. Walter, and A. Wilson, "Real-time dynamic loading and thermal diagnostic of power transformers," *IEEE*
- dynamic loading and thermal diagnostic of power transformers," *IEEE Trans. Power Del.*, vol. 18, no. 1, pp. 142–148, Jan. 2003.
 [241] S. A. Ryder and I. J. Vaughan, "A simple method for calculating core
- temperature rise in power transformers," IEEE Trans. Power Del., vol. 19, no. 2, pp. 637–642, Apr. 2004. [242] G. Swift, T. S. Molinski, and W. Lehn, "A fundamental approach to
- transformer thermal modeling-Part I: Theory and equivalent circuit," *IEEE Trans. Power Del.*, vol. 16, no. 2, pp. 171–175, Apr. 2001. [243] G. Swift, T. S. Molinski, R. Bray, and R. Menzies, "A fundamental
- approach to transformer thermal modeling-Part II: Field verification," IEEE Trans. Power Del., vol. 16, no. 2, pp. 176-180, Apr. 2001.
- [244] Z. Radakovic and K. Feser, "A new method for the calculation of the hot-spot temperature in power transformers with ONAN cooling, IEEE Trans. Power Del., vol. 18, no. 4, pp. 1284-1292, Oct. 2003.

- [245] W. H. Tang, Q. H. Wu, and Z. J. Richardson, "A simplified transformer thermal model based on thermal-electric analogy," IEEE Trans. Power Del., vol. 19, no. 3, pp. 1112-1119, Jul. 2004.
- [246] V. Galdi, L. Ippolito, A. Piccolo, and A. Vaccaro, "Parameter identification of power transformers thermal model via genetic algorithms," Electr. Power Syst. Res., vol. 60, pp. 107-113, 2001.
- [247] D. Susa, M. Lehtonen, and H. Nordman, "Dynamic thermal modelling of power transformers," IEEE Trans. Power Del., vol. 20, no. 1, pp. 197-204, Jan. 2005.
- [248] D. Susa and M. Lehtonen, "Dynamic thermal modeling of power transformers: Further development-Part I," IEEE Trans. Power Del., vol. 21, no. 4, pp. 1961–1970, Oct. 2006.
- [249] D. Susa and M. Lehtonen, "Dynamic thermal modeling of power transformers: Further development-Part II," IEEE Trans. Power Del., vol. 21, no. 4, pp. 1971-1980, Oct. 2006.
- [250] X. Mao, D. J. Tylavsky, and G. A. McCulla, "Assessing the reliability of linear dynamic transformer thermal modelling," *Proc. Inst. Elect. Eng., Gen. Transm. Distrib.*, vol. 153, no. 4, pp. 414-422, Jul. 2006. [251] J. Zhang and X. Li, "Coolant flow distribution and pressure loss in
- ONAN transformer windings-Part I: Theory and model development," IEEE Trans. Power Del., vol. 19, no. 1, pp. 186-193, Jan. 2004.
- [252] J. Zhang and X. Li, "Coolant flow distribution and pressure loss in ONAN transformer windings-Part II: Optimization of design param-
- [253] J. Zhang and X. Li, "Oil cooling for disk-type transformer wind-ings—Part 1: Theory and model development," *IEEE Trans. Power* Del., vol. 21, no. 3, pp. 1318–1325, Jul. 2006. [254] J. Zhang and X. Li, "Oil cooling for disk-type transformer wind-
- ings-Part II: Parametric studies of design parameters," IEEE Trans. Power Del., vol. 21, no. 3, pp. 1326-1332, Jul. 2006.
- [255] S. W. Heunis and R. Herman, "A thermal loading guide for residential distribution transformers based on time-variant current load models," *IEEE Trans. Power Syst.*, vol. 19, no. 3, pp. 1294–1298, Aug. 2004. [256] X. Li, R. W. Mazur, D. R. Allen, and D. R. Swatek, "Specifying trans-
- former winter and summer peak-load limits," IEEE Trans. Power Del., vol. 20, no. 1, pp. 185–190, Jan. 2005. [257] K. C. Schneider and R. F. Hoad, "Initial transformer sizing for single
- phase residential load," IEEE Trans. Power Del., vol. 7, no. 4, pp. 2074–2081, Oct. 1992.
- [258] W. Fu, J. D. McCalley, and V. Vittal, "Risk assessment for transformer loading," IEEE Trans. Power Syst., vol. 16, no. 3, pp. 346-353, Aug. 2001.
- [259] T. T. Nguyen, "Constrained optimisation procedure for evaluating cyclic loading of power transformers," Proc. Inst. Elect Eng., Gen. *Transm. Distrib.*, vol. 142, no. 3, pp. 240–246, May 1995. [260] W. H. Tang, E. Prempain, Q. H. Wu, and J. Fitch, "A particle swarm
- optimizer with passive congregation approach to thermal modelling for power transformers," in Proc. IEEE Congress on Evolutionary Computation, Sep. 2005, vol. 3, pp. 2745-2751.
- [261] Q. He, J. Ši, and D. J. Tylavsky, "Prediction of top-oil temperature for transformers using neural networks," IEEE Trans. Power Del., vol. 15, no. 4, pp. 1205-1211, Oct. 2000.
- [262] M. Hell, P. Costa, Jr., and F. Gomide, "Recurrent neurofuzzy network in thermal modeling of power transformers," *IEEE Trans. Power Del.*, vol. 22, no. 2, pp. 904–910, Apr. 2007.
- [263] M. Hell, P. Costa, and F. Gomide, "Participatory learning in power transformers thermal modeling," IEEE Trans. Power Del., vol. 23, no. 4, pp. 2058-2067, Oct. 2008.
- [264] D. Peterchuck and A. Pahwa, "Sensitivity of transformer's hottest-spot and equivalent aging to selected parameters," IEEE Trans. Power Del., vol. 17, no. 4, pp. 996–1001, Oct. 2002. [265] L. Pierce, "An investigation of the temperature distribution in cast-
- resin transformer windings," IEEE Trans. Power Del., vol. 7, no. 2, pp. 920-926, Apr. 1992.
- [266] L. Pierce, "An investigation of the thermal performance of an oil filled transformer," IEEE Trans. Power Del., vol. 7, no. 3, pp. 1347-1358, Jul. 1992.
- [267] L. Pierce, "Predicting liquid filled transformer loading capability,"
- *IEEE Trans. Ind. Appl.*, vol. 30, no. 1, pp. 170–178, Jan. 1994. [268] D. J. Tylavsky, X. Mao, and G. A. McCulla, "Transformer thermal modeling: Improving reliability using data quality control," IEEE Trans. Power Del., vol. 21, no. 3, pp. 1357-1366, Jul. 2006.
- [269] M. Ahamed, A. Wahab, M. Mahmoud, H. Assad, G. Zeitoun, and G. Ismail, "A newly modified forced oil cooling system and its impact on in-service transformer oil characteristics," IEEE Trans. Power Del., vol. 18, no. 3, pp. 827-834, Jul. 2003.
- [270] P. Picher, L. Bolduc, A. Dutil, and V. Q. Pham, "Study of the acceptable DC current limit in core-form power transformers," *IEEE Trans. Power Del.*, vol. 12, no. 1, pp. 257–265, Jan. 1997.
- [271] O. Bíró, S. Außerhofer, G. Buchgraber, K. Preis, and W. Seitlinger, "Prediction of magnetizing current waveform in a single-phase power transformer under DC bias," *Inst. Eng. Technol. Sci. Meas. Tech.*, vol. 1, no. 1, pp. 2-5, Jan. 2007.

- [272] O. Biro, G. Buchgraber, G. Leber, and K. Preis, "Prediction of magnetizing current wave-forms in a three-phase power transformer under DC bias," IEEE Trans. Magn., vol. 44, no. 6, pp. 1554-1557, Jun. 2008.
- [273] S. Lu and Y. Liu, "FEM analysis of DC saturation to assess transformer susceptibility to geomagnetically induced currents," *IEEE Trans. Power Del.*, vol. 8, no. 3, pp. 1367–1376, Jul. 1993. [274] S. Lu, Y. Liu, and J. De La Ree, "Harmonics generated from a DC bi-
- ased transformer," IEEE Trans. Power Del., vol. 8, no. 2, pp. 725-731, Apr. 1993.
- [275] W. C. Viana, R. J. Micaleff, S. Young, and F. P. Dawson, "Transformer design considerations for mitigating geomagnetic induced saturation,' *IEEE Trans. Magn.*, vol. 35, no. 5, pp. 3532–3534, Sep. 1999. [276] E. F. Fuchs, Y. You, and D. J. Roesler, "Modeling and simulation, and
- their validation of three-phase transformers with three legs under DC bias," IEEE Trans. Power Del., vol. 14, no. 2, pp. 443-449, Apr. 1999.
- [277] P. R. Price, "Geomagnetically induced current effects on transformers," IEEE Trans. Power Del., vol. 17, no. 4, pp. 1002-1008, Oct. 2002.
- [278] W. Chandrasena, P. G. McLaren, U. D. Annakkage, and R. P. Jayasinghe, "An improved low-frequency transformer model for use in GIC studies," IEEE Trans. Power Del., vol. 19, no. 2, pp. 643-651, Apr. 2004.
- [279] Y. Yao, C. S. Koh, G. Ni, and D. Xie, "3-D nonlinear transient eddy current calculation of online power transformer under DC bias," IEEE *Trans. Magn.*, vol. 41, no. 5, pp. 1840–1843, May 2005. [280] A. Ahfock and A. J. Hewitt, "DC magnetization of transformers," *Proc.*
- Inst. Elect. Eng., Electr. Power Appl., vol. 153, no. 4, pp. 601-607, Jul. 2006
- [281] M. Lahtinen and J. Elovaara, "GIC occurrences and GIC test for 400 kV system transformer," IEEE Trans. Power Del., vol. 17, no. 2, pp. 555-561, Apr. 2002.
- [282] A. Medina, G. Saucedo, C. M. Sánchez, A. M. Maldonado, and L. Sánchez, "Experimental determination of saturation characteristics in multilimb power transformers using DC excitation," IEEE Power Eng. Rev., vol. 22, no. 11, pp. 47-48, Nov. 2002.
- [283] Y. You, E. F. Fuchs, D. Lin, and P. R. Barnes, "Reactive power demand of transformers with DC bias," IEEE Ind. Appl. Mag., vol. 2, no. 4, pp. 45–52, Jul./Aug. 1996. [284] S. B. Williams, P. A. Abetti, and E. F. Magnesson, "How digital com-
- puters aid transformer designer," in Gen. Elect. Rev.. Schenectady, NY: , May 1955, vol. 58, pp. 24-25.
- [285] S. B. Williams, P. A. Abetti, and E. F. Magnusson, "Application of digital computers to transformer design," AIEE Trans. Power App. Syst., vol. 75, pt. III, pp. 728–735, 1956.
- W. Sharpley and J. V. Oldfield, "The digital computer applied to the [286] design of large power transformers," Proc. Inst. Elect. Eng., vol. 105, pt. A, pp. 112–125, 1958. [287] S. B. Williams, P. A. Abetti, and H. J. Mason, "Complete design of
- power transformers with a large size digital computer," AIEE Trans. Power App. Syst., vol. 77, pt. III, pp. 1282-1291, 1959.
- [288] O. W. Andersen, "Optimum design of electrical machines," IEEE Trans. Power App. Sys., vol. PAS-86, no. 6, pp. 707-711, Jun. 1967.
- [289] H. H. Wu and R. Adams, "Transformer design using time-sharing computer," IEEE Trans. Magn., vol. MAG-6, no. 1, p. 67, Mar. 1970.
- [290] P. Odessey, "Transformer design by computer," IEEE Trans. Manuf. Technol., vol. 3, no. 1, pp. 1-17, Jan. 1974
- [291] F. F. Judd and D. R. Kressler, "Design optimization of small low-frequency power transformer," IEEE Trans. Magn., vol. MAG-13, no. 4, pp. 1058–1069, Jul. 1977.
- W. G. Hurley, W. H. Wolfle, and J. G. Breslin, "Optimized transformer design: Inclusive of high-frequency effects," *IEEE Trans. Power Elec-*[292] tron., vol. 13, no. 4, pp. 651-659, Jul. 1998.
- [293] M. Poloujadoff and R. D. Findlay, "A procedure for illustrating the effect variations of parameters on optimal transformers design," IEEE Trans. Power Syst., vol. PWRS-1, no. 4, pp. 202–205, Nov. 1986. W. T. Jewell, "Transformer design in the undergraduate power engi-
- [294] neering laboratory," IEEE Trans. Power Sys., vol. 5, no. 2, pp. 499-505, May 1990.
- [295] W. M. Grady, R. Chan, M. J. Samotyj, R. J. Ferraro, and J. L. Bierschenk, "A PC-based computer program for teaching the design and analysis of dry-type transformers," *IEEE Trans. Power Syst.*, vol. 7, no. 2, pp. 709-717, May 1992.
- [296] A. Rubaai, "Computer aided instruction of power transformer design in the undergraduate power engineering class," IEEE Trans. Power Syst., vol. 9, no. 3, pp. 1174–1181, Aug. 1994. [297] O. W. Andersen, "Optimized design of electric power equipment,"
- IEEE Comput. Appl. Power, vol. 4, no. 1, pp. 11-15, Jan. 1991.
- [298] C. Hernandez, M. A. Arjona, and S.-H. Dong, "Object-oriented knowledge-based system for distribution transformer design," IEEE Trans. Magn., vol. 44, no. 10, pp. 2332–2337, Oct. 2008.
- [299] R. A. Jabr, "Application of geometric programming to transformer design," IEEE Trans. Magn., vol. 41, no. 11, pp. 4261-4269, Nov. 2005.

- [300] E. I. Amoiralis, P. S. Georgilakis, M. A. Tsili, and A. G. Kladas, "Global transformer optimization method using evolutionary design and numerical field computation," *IEEE Trans. Magn.*, vol. 45, no. 3, pp. 1720-1723, Mar. 2009.
- [301] C. J. Wu and F. C. Lee, "Minimum weight EI core and pot core inductor and transformer designs," IEEE Trans. Magn., vol. 16, no. 5, pp. 755-757, Sep. 1980.
- [302] N. D. Doulamis, A. D. Doulamis, P. S. Georgilakis, S. D. Kollias, and N. D. Hatziargyriou, "A synergetic neural network-genetic scheme for optimal transformer construction," Integr. Comput.-Aided Eng., vol. 9, pp. 37-56, 2002.
- [303] N. D. Doulamis and A. D. Doulamis, "Optimal distribution transformers assembly using an adaptable neural network-genetic algorithm scheme," in Proc. IEEE Int. Conf. Systems, Man Cybernetics, Oct. 2002, vol. 5, pp. 6. [304] T. H. Pham, S. J. Salon, and S. R. H. Hoole, "Shape optimization of
- windings for minimum losses," IEEE Trans. Magn., vol. 32, no. 5, pp. 4287-4289, Sept. 1996.
- [305] P. S. Georgilakis, M. A. Tsili, and A. T. Souflaris, "A heuristic solution to the transformer manufacturing cost optimization problem," J. Mat.
- Process. Technol., vol. 181, pp. 260–266, 2007.
 [306] L. Hui, H. Li, H. Bei, and Y. Shunchang, "Application research based on improved genetic algorithm for optimum design of power transformers," in Proc. 5th Int. Conf. Electrical Machines and Systems, 2001, vol. 1, pp. 242-245.
- [307] S. Elia, G. Fabbri, E. Nistico, and E. Santini, "Design of cast-resin distribution transformers by means of genetic algorithms," in Proc. Int. Symp. Power Electronics, Electrical Drives, Automation and Motion, 2006, pp. 1473–1477.
- [308] N. Tutkun and A. Moses, "Design optimization of a typical strip-wound toroidal core using genetic algorithms," J. Magn. Magn. Mater., vol. 277, no. 1-2, pp. 216-220, 2004.
- [309] E. I. Amoiralis, P. S. Georgilakis, T. D. Kefalas, M. A. Tsili, and A. G. Kladas, "Artificial intelligence combined with hybrid FEM-BE techniques for global transformer optimization," IEEE Trans. Magn., vol. 43, no. 4, pp. 1633–1636, Apr. 2007. [310] L. H. Geromel and C. R. Souza, "The applications of intelligent sys-
- tems in power transformer design," in Proc. IEEE Canadian Conf. Electrical and Computer Engineering, 2002, vol. 1, pp. 285-290.
- [311] N. Takahashi, T. Kitamura, M. Horii, and J. Takehara, "Optimal design of tank shield model of transformer," IEEE Trans. Magn., vol. 36, no. 4, pp. 1089-1093, Jul. 2000.
- [312] M. A. Tsili, A. G. Kladas, P. S. Georgilakis, A. T. Souflaris, and D. G. Paparigas, "Geometry optimization of magnetic shunts in power transformers based on a particular hybrid finite-element boundary-element model and sensitivity analysis," IEEE Trans. Magn., vol. 41, no. 5, pp. 1776-1779, May 2005.
- [313] J. C. Olivares, Y. L. Jose, M. Canedo, R. Escarela-Pérez, J. Driesen, and P. Moreno, "Reducing losses in distribution transformers," IEEE Trans. Power Del., vol. 18, no. 3, pp. 821-826, Jul. 2003.
- [314] M. Rizzo, A. Savini, and J. Turowski, "Dependence of forces, eddy current and stray losses on screening in power transformers," in Proc. 5th Int. Conf. Electrical Machines and Systems, Aug. 2001, vol. 1, pp. 182 - 185.
- [315] B. Kennedy, Energy Efficient Transformers. New York: Mc-Graw-Hill, 1998.
- [316] J. F. Baranowski, T. A. Edison, and P. J. Hopkinson, "An alternative evaluation of distribution transformers to achieve the lowest TOC,"
- IEEE Trans. Power Del., vol. 7, no. 2, pp. 614–619, Apr. 1992.
 [317] W. T. J. Hulshorst and J. F. Groeman, "Energy saving in industrial distribution transformers," 2002, KEMA Rep..
- [318] S. Y. Merritt and S. D. Chaitkin, "No load versus load losses," IEEE *Ind. Appl. Mag.*, vol. 9, no. 6, pp. 21–28, Nov. 2003. [319] S. Y. Merritt and S. D. Chaitkin, "One from menu A—One from menu
- B," *IEEE Ind. Appl. Mag.*, vol. 11, no. 4, pp. 66–71, Jul. 2005. [320] P. S. Georgilakis, "Decision support system for evaluating transformer investments in the industrial sector," J. Mat. Process. Technol., vol. 181, pp. 307-312, 2007.
- [321] D. L. Nickel and H. R. Braunstein, "Distribution transformer loss eval-uation: I—Proposed techniques," *IEEE Trans. Power App. Sys.*, vol. PAS-100, no. 2, pp. 788-797, Feb. 1981.
- [322] D. L. Nickel and H. R. Braunstein, "Distribution transformer loss evaluation: II-Load characteristics and system cost parameters," IEEE *Trans. Power App. Syst.*, vol. 100, no. 2, pp. 798–811, Feb. 1981. [323] B. W. McConnell, "Increasing distribution transformer efficiency: Po-
- tential for energy savings," Proc. IEEE Power Eng. Rev., vol. 18, no. 7, pp. 8–10, 1998.
- [324] "External costs: Research results on socio-environmental damages due to electricity and transport, 2003, Eur. Comm., 2003, Brussels: Directorate-General for Research, Study 20198
- [325] M. Vesterdal and G. T. Svendsen, "How should greenhouse gas permits be allocated in the EU?," Energy Policy, vol. 32, pp. 961-968, 2004.

- [326] R. Targosz, R. Belmans, J. Declercq, H. De Keulenaer, K. Furuya, M. Karmarkar, M. Martinez, M. McDermott, and I. Pinkiewicz, "The potential for global energy savings from high energy efficiency distribution transformers," 2005, Leonardo Energy, Eur. Copper Inst.
- [327] E. Delarue, H. Lamberts, and W. D'haeseleer, "Simulating greenhouse gas (GHG) allowance cost and GHG emission reduction in Western Europe," Energy, vol. 32, pp. 1299-1309, 2007.
- [328] J. Frau, J. Gutierrez, and A. Ramis, "Consider the true cost of transformer losses," Transm. Distrib. World, vol. 1, pp. 50-55, 2007
- [329] J. L. Goudie and W. J. Chatterton, "Environmental and life cycle considerations for distribution and small power transformer selection and specification," in Proc. IEEE Int. Symp. Electrical Insulation, 2002, pp. 354-358
- [330] T. Stensland, E. F. Fuchs, W. M. Grady, and M. T. Doyle, "Modeling of magnetizing and core-loss currents in single-phase transformers with voltage harmonics for use in power flow," IEEE Trans. Power Del., vol. 12, no. 2, pp. 768–774, Apr. 1997. [331] J. Pedra, F. Córcoles, L. Sainz, and R. López, "Harmonic nonlinear
- transformer modeling," IEEE Trans. Power Del., vol. 19, no. 2, pp. 884-890, Apr. 2004.
- [332] O. A. Mohammed, N. Y. Abed, and S. Liu, "Investigation of the harmonic behavior of three phase transformer under nonsinusoidal operation using finite element and wavelet packets," IEEE Trans. Magn., vol. 42, no. 4, pp. 967-970, Apr. 2006.
- [333] A. H. Chowdhury, W. M. Grady, and E. F. Fuchs, "An investigation of the harmonic characteristics of transformer excitation current under nonsinusoidal supply voltage," IEEE Trans. Power Del., vol. 14, no. 2, pp. 450-458, Apr. 1999.
- [334] M. A. S. Masoum, P. S. Moses, and A. S. Masoum, "Derating of asymmetric three-phase transformers serving unbalanced nonlinear loads," *IEEE Trans. Power Del.*, vol. 23, no. 4, pp. 2033–2041, Oct. 2008. [335] J. A. Martinez and B. A. Mork, "Transformer modeling for low- and
- mid-frequency transients-A review," IEEE Trans. Power Del., vol. 20, no. 2, pt. 2, pp. 1625–1632, Apr. 2005. [336] J. A. Martinez, R. Walling, B. A. Mork, J. Martin-Arnedo, and
- D. Durbak, "Parameter determination for modeling system transients-Part III: Transformers," IEEE Trans. Power Del., vol. 20, no. 3, pp. 2051–2062, Jul. 2005.
- [337] V. Woivre, J. P. Arthaud, A. Ahmad, and N. Burais, "Transient (5) Y. Wolte, J. T. Arlhad, A. Amhad, and N. Buras, Transferr overvoltage study and model for shell-type power transformers," *IEEE Trans. Power Del.*, vol. 8, no. 1, pp. 212–222, Jan. 1993.
 [338] M. Gutierrez and R. C. Degeneff, "Linear, lumped parameter trans-
- former model reduction technique," *IEEE Trans. Power Del.*, vol. 10, no. 2, pp. 853–861, Apr. 1995. [339] R. C. Degeneff, M. Gutierrez, and M. Vakilian, "Nonlinear, lumped
- parameter transformer model reduction technique," IEEE Trans. Power *Del.*, vol. 10, no. 2, pp. 862–868, Apr. 1995. F. de Leon and A. Semlyen, "A simple representation of dynamic hys-
- [340] teresis losses in power transformers," IEEE Trans. Power Del., vol. 10, no. 1, pp. 315-321, Jan. 1995.
- [341] F. de Leon and A. Semlyen, "Efficient calculation of elementary parameters of transformers," IEEE Trans. Power Del., vol. 7, no. 1, pp. 376-383, Jan. 1992.
- [342] F. de Leon and A. Semlyen, "Reduced order model for transformer transients," IEEE Trans. Power Del., vol. 7, no. 1, pp. 361-369, Jan. 1992.
- [343] F. de Leon and A. Semlyen, "Time domain modelling of eddy current effects for transformer transients," IEEE Trans. Power Del., vol. 8, no. 1, pp. 271-279, Jan. 1993.
- [344] F. de Leon and A. Semlyen, "Computation of electro-magnetic transients using dual or multiple time steps," IEEE Trans. Power Syst., vol.
- 8, no. 3, pp. 1274–1281, Aug. 1993.
 [345] F. de Leon and A. Semlyen, "Complete transformer model for electromagnetic transients," *IEEE Trans. Power Del.*, vol. 9, no. 1, pp. 231-239, Jan. 1994.
- [346] F. de Leon and A. Semlyen, "Detailed modeling of eddy current effects for transformer transients," *IEEE Trans. Power Del.*, vol. 9, no. 2, pp. 1143-1150, Apr. 1994.
- [347] A. Narang and R. H. Brierley, "Topology based magnetic model for steady-state and transient studies for three-phase core type transformers," IEEE Trans. Power Syst., vol. 9, no. 3, pp. 1337-1349, Aug. 1994
- [348] X. Chen and S. S. Venkata, "A three-phase three-winding core-type transformer model for low-frequency transient studies," IEEE Trans. *Power Del.*, vol. 12, no. 2, pp. 775–782, Apr. 1997.
 [349] B. C. Papadias, N. D. Hatziargyriou, J. A. Bakopoulos, and J. M.
- Prousalidis, "Three phase transformer modelling for fast electromag-netic transient studies," *IEEE Trans. Power Del.*, vol. 9, no. 2, pp. 1151-1159, Apr. 1994.
- [350] L. Haydock and S. A. Holland, "Transient analysis of power transformers using magnetic and equivalent circuits," IEEE Trans. Magn., vol. 30, no. 5, pp. 2996-2999, Sep. 1994.

- [351] A. Tokic, V. Madzarevic, and I. Uglesic, "Numerical calculations of three-phase transformer transients," IEEE Trans. Power Del., vol. 20, no. 4, pp. 2493-2500, Oct. 2005.
- [352] A. Tokic and I. Uglesic, "Elimination of overshooting effects and suppression of numerical oscillations in transformer transient calculations," IEEE Trans. Power Del., vol. 23, no. 1, pp. 243-251, Jan. 2008.
- [353] W. Hribernik, L. Graber, and J. H. Brunke, "Inherent transient recovery voltage of power transformers—A model-based determination procedure,⁵⁷ *IEEE Trans. Power Del.*, vol. 21, no. 1, pp. 129–134, Jan. 2006. [354] K. G. N. B. Abeywickrama, A. D. Podoltsev, Y. V. Serdyuk, and S. M.
- Gubanski, "Computation of parameters of power transformer windings for use in frequency response analysis," IEEE Trans. Magn., vol. 43, no. 5, pp. 1983–1990, May 2007. [355] N. Abeywickrama, Y. V. Serdyuk, and S. M. Gubanski, "High-fre-
- quency modeling of power transformers for use in frequency response analysis (FRA)," *IEEE Trans. Power Del.*, vol. 23, no. 4, pp. 2042–2049, Oct. 2008.
- [356] D. J. Wilcox, "Theory of transformer modelling using modal anal-ysis," Proc. Inst. Elect. Eng., Gen. Transm. Distrib., vol. 138, no. 2, pp. 121–128, Mar. 1991
- S. Song, H. S. Kim, T. K. Chung, and S. Y. Hahn, "Modal analysis [357] for the transient internal voltage of a transformer," IEEE Trans. Magn.,
- vol. 40, no. 2, pp. 581–584, Mar. 2004.
 [358] H. Mohseni, "Multi-winding multi-phase transformer model with saturable core," *IEEE Trans. Power Del.*, vol. 6, no. 1, pp. 166–173, Jan. 1991.
- [359] R. M. Del Vecchio, "Applications of a multiterminal transformer model using two winding leakage inductances," IEEE Trans. Power Del., vol. 21, no. 3, pp. 1300-1308, Jul. 2006.
- [360] R. M. D. Vecchio, "Multiterminal three phase transformer model with balanced or unbalanced loading," IEEE Trans. Power Del., vol. 23, no. 3, pp. 1439–1447, Jul. 2008.
- [361] D. Dolinar, J. Pihler, and B. Grcar, "Dynamic model of a three-phase power transformer," *IEEE Trans. Power Del.*, vol. 8, no. 4, pp. 1011–1010, oc. (2007). 1811-1819, Oct. 1993.
- [362] C. Koley, P. Purkait, and S. Chakravorti, "Time-frequency representation of resistance for modeling of transformer winding under impulse test," IEEE Trans. Power Del., vol. 21, no. 3, pp. 1367-1374, Jul, 2006.
- [363] S. Arunkumar, V. Sandeep, S. Shankar, M. Gopalakrishnan, K. Udayakumar, and V. Jayashankar, "Impulse testing of power transformers—A model reference approach," *Proc. Inst. Elect. Eng., Sci. Meas. Technol.*, vol. 151, no. 1, pp. 25–30, Jun. 2004.
 [364] D. L. Stuehm, B. A. Mork, and D. D. Mairs, "Five-legged core transformer equivalent circuit," *IEEE Trans. Power Del.*, vol. 4, no. 3, pp. 1766–1702. Jul 1090.
- 1786–1793, Jul. 1989.
- [365] B. A. Mork, "Five-legged wound-core transformer model: Derivation, parameters, implementation and evaluation," IEEE Trans. Power Del., vol. 14, no. 4, pp. 1519–1526, Oct. 1999. [366] B. A. Mork, F. Gonzalez, D. Ishchenko, D. L. Stuehm, and J. Mitra,
- "Hybrid transformer model for transient simulation-Part I: Development and parameters," IEEE Trans. Power Del., vol. 22, no. 1, pp. 248-255, Jun. 2007.
- [367] B. A. Mork, F. Gonzalez, D. Ishchenko, D. L. Stuehm, and J. Mitra, "Hybrid transformer model for transient simulation-Part II: Laboratory measurements and benchmarking," IEEE Trans. Power Del., vol. 22, no. 1, pp. 256-262, Jun. 2007.
- [368] B. Mork, D. Ishchenko, F. Gonzalez, and S. D. Cho, "Parameter estimation methods for five-limb magnetic core model," IEEE Trans. Power Del., vol. 23, no. 4, pp. 2025–2032, Oct. 2008. [369] S. Fujita, N. Hosokawa, and Y. Shibuya, "Experimental investigation
- of high frequency voltage oscillation in transformer windings," IEEE *Trans. Power Del.*, vol. 13, no. 4, pp. 1201–1207, Oct. 1998. [370] M. Popov, L. van der Sluis, G. C. Paap, and H. De Herdt, "Computa-
- tion of very fast transient overvoltages in transformer windings," IEEE Trans. Power Del., vol. 18, no. 4, pp. 1268-1274, Oct. 2003
- [371] M. Popov, L. van der Sluis, R. Peter, P. Smeets, and J. L. Roldan, "Analysis of very fast transients in layer-type transformer windings," IEEE *Trans. Power Del.*, vol. 22, no. 1, pp. 238–247, Jun. 2007. [372] S. Fujita, Y. Shibuya, and M. Ishii, "Influence of VFT on shell-type
- transformer," IEEE Trans. Power Del., vol. 22, no. 1, pp. 217-222, Jan. 2007
- [373] S. M. H. Hosseini, M. Vakilian, and G. B. Gharehpetian, "Comparison of transformer detailed models for fast and very fast transient studies, IEEE Trans. Power Del., vol. 23, no. 2, pp. 733-741, Apr. 2008
- [374] E. E. Mombello and K. Möller, "New power transformer model for the calculation of electromagnetic resonant transient phenomena including frequency-dependent losses," *IEEE Trans. Power Del.*, vol. 15, no. 1, pp. 167–174, Jan. 2000.
- [375] E. E. Mombello, "Impedances for the calculation of electromagnetic transients within transformers," IEEE Trans. Power Del., vol. 17, no. pp. 479-488, Apr. 2002.
- [376] T. Noda, H. Nakamoto, and S. Yokoyama, "Accurate modeling of coretype distribution transformers for electromagnetic transient studies," IEEE Trans. Power Del., vol. 17, no. 4, pp. 969-976, Oct. 2002.

- [377] J. Leohold and P. P. Silvester, "Calculation of transformer winding resonances by a specialized finite element method employing substructures," IEEE Trans. Power Syst., vol. 3, no. 3, pp. 1385-1389, Aug. 1988
- [378] O. A. Mohammed, Z. Liu, S. Liu, and N. Y. Abed, "Finite-elementbased nonlinear physical model of iron-core transformers for dynamic simulations," IEEE Trans. Magn., vol. 42, no. 4, pp. 1027-1030, Apr. 2006.
- [379] Z. Liu, S. Liu, and O. A. Mohammed, "A practical method for building the FE-based phase variable model of single phase transformers for dynamic simulations," IEEE Trans. Magn., vol. 43, no. 4, pp. 1761-1764, Apr. 2007.
- [380] B. Gustavsen, "Wide band modeling of power transformers," IEEE Trans. Power Del., vol. 19, no. 1, pp. 414-422, Jan. 2004.
- [381] J. Pedra, L. Sainz, F. Córcoles, R. López, and M. Salichs, "PSpice computer model of a nonlinear three-phase three-legged transformer," IEEE *Trans. Power Del.*, vol. 19, no. 1, pp. 200–207, Jan. 2004. [382] L. Guasch, F. Corcoles, J. Pedra, and L. Sainz, "Effects of symmetrical
- voltage sags on three-phase three-legged transformers," IEEE Trans. *Power Del.*, vol. 19, no. 2, pp. 875–883, Apr. 2004. [383] J. Pedra, L. Sainz, F. Corcoles, and L. Guasch, "Symmetrical and un-
- symmetrical voltage sag effects on three-phase transformers," *IEEE Trans. Power Del.*, vol. 20, no. 2, pt. 2, pp. 1683–1691, Apr. 2005.
 [384] S. Sieradzki, R. Rygal, and M. Soinski, "Apparent core losses and core
- losses in five-limb amorphous transformer of 160 kVA," IEEE Trans. Magn., vol. 34, no. 4, pp. 1189-1191, Jul. 1998.
- [385] M. Yamamoto, T. Mori, T. Kawasaki, K. Tsutsui, T. Itoh, and T. Yasigawa, "A design study of amorphous core transformer," IEEE Trans. Magn., vol. 20, no. 5, pp. 1771-1773, Sep. 1984.
- [386] A. Coombs, M. Lindenmo, D. Snell, and D. Power, "Review of the types, properties, advantages, and latest developments in insulating coatings on non-oriented electrical steels," IEEE Trans. Magn., vol. 37, no. 1, pp. 544-557, Jan. 2001.
- [387] T. Matsuura, K. Naqayama, S. Hagiwara, and M. Higaki, "Long-term reliability of iron-based amorphous alloy cores for oil-immersed transformers," IEEE Trans. Magn., vol. 26, no. 5, pp. 1993–1995, Sep. 1990.
- [388] S. Ezure, Y. Imai, H. Sato, S. Yamada, K. Yamanaka, and S. Saito, "Long-term reliability of amorphous alloy wound core distribution transformers," IEEE Trans. Power Del., vol. 9, no. 1, pp. 249-256, Jan. 1994.
- [389] T. Nitta, K. Misawa, and H. Nomura, "Some considerations on superconducting transformers from a design-point of view," *IEEE Trans. Magn.*, vol. 32, no. 4, pp. 2381–2384, Jul. 1996.
- [390] E. N. Andreev and Li. I. Chubraeva, "Investigation of a model HTSC transformer with amorphous alloy cores," J. Mat. Process. Tech., vol. 181, no. 1-3, pp. 25-30, Jan. 2007.
- [391] M. Ariante, A. Formisano, F. Marignetti, R. Martone, G. Masullo, A. Matrone, R. Quarantiello, G. Rubinacci, F. Sangiorgi, M. Scarano, S. Silvestri, F. Villone, and M. Zigon, "New perspectives in HTS transformer design," in J. Phys.: Conf. Ser., 2006, vol. 43, pp. 845-848.
- [392] J. K. Sykulski, C. Beduz, R. L. Stoll, M. R. Harris, K. F. Goddard, and Y. Yang, "Prospects for large high-temperature superconducting power transformers: Conclusions from a design study," *Proc. Inst. Elect. Eng., Electr. Power Appl.*, vol. 146, no. 1, pp. 41–52, 1999.
- [393] J.-M. Joung, S.-M. Baek, and S.-H. Kim, "Manufacturing and test of model double-pancake coils of HTS transformer for cryogenic insulation design," IEEE Trans. Appl. Supercond., vol. 14, no. 2, pp. 928-931, Jun. 2004.
- [394] J.-K. Lee, W.-S. Kim, and S.-Y. Hahn, "Development of a three phase 100 kVA superconducting power transformer with amorphous cores, IEEE Trans. Appl. Supercond., vol. 9, no. 2, pp. 1293-1296, Jun. 1999.
- [395] H. Kamijo, H. Hata, H. Fujimoto, K. Ikeda, T. Herai, K. Sakaki, H. Yamada, Y. Sanuki, S. Yoshida, Y. Kamioka, M. Iwakuma, and K. Funaki, "Fabrication of inner secondary winding of high-TC superconducting traction transformer for railway rolling stock," *IEEE Trans. Appl. Su*percond., vol. 15, no. 2, pp. 1875-1878, Jun. 2005.
- [396] K. Fukuoka and M. Hashimoto, "Proposal of transformer using magnetic shielding with bulk high Tc superconductors," IEEE Trans. Appl. *Supercond.*, vol. 14, no. 2, pp. 1126–1129, Jun. 2004. [397] J.-T. Kim, W.-S. Kim, S.-H. Kim, K.-D. Choi, J.-H. Han, G.-W. Hong,
- and S.-Y. Hahn, "Analysis of AC losses in HTS pancake windings for transformer according to the operating temperature," IEEE Trans. Magn., vol. 41, no. 5, pp. 1888-1891, May 2005.
- [398] T. Ö. Rouse, "Mineral insulating oil in transformers," IEEE Electr.
- *Insul. Mag.*, vol. 14, no. 3, pp. 6–16, May 1998. [399] Y. Bertrand and L. C. Hoang, "Vegetal oils as substitute for mineral oils," in Proc. Int. Conf. Properties and Applications of Dielectric Materials, Jun. 2003, vol. 2, pp. 491-494.
- [400] C. Sun, J. Li, X. Li, and S. Grzybowski, "Electric properties of vegetable oil-based dielectric liquid and lifetime estimation of the oil-paper insulation," in Proc. IEEE Conf. Electrical Insulation and Dielectric Phenomena, Oct. 2006, pp. 680-683.

- [401] D. Martin, Z. D. Wang, A. W. Darwin, and I. James, "A comparative study of the chemical stability of esters for use in large power transformers," in *Proc. IEEE Conf. Electrical Insulation and Dielectric Phenomena*, Oct. 2006, pp. 493–496.
- [402] C. P. McShane, "Vegetable-oil-based dielectric coolants," *IEEE Ind. Appl. Mag.*, vol. 8, no. 3, pp. 34–41, May/Jun. 2002.
- [403] T. V. Oommen, "Vegetable oils for liquid-filled transformers," *IEEE Electr. Insul. Mag.*, vol. 8, no. 1, pp. 6–11, Jan./Feb. 2002.
- [404] E. A. Feilat, S. Grzybowski, and P. Knight, "Accelerated aging of high voltage encapsulated transformers for electronics applications," in Proc. 6th Int. Conf. Properties and Applications of Dielectric Materials, 2000, vol. 1, pp. 209–212.
- [405] C. P. McShane, K. J. Rapp, J. L. Corkran, G. A. Gauger, and J. Luksich, "Aging of paper insulation in natural ester dielectric fluid," in *Proc. IEEE/Power Eng. Soc. Transmission Distribution Conf. Expo.*, 2001, vol. 2, pp. 675–679.
- [406] C. Perrier, A. Beroual, and J.-L. Bessede, "Improvement of power transformers by using mixtures of mineral oil with synthetic esters," *IEEE Trans. Dielectr. Electr. Insul.*, vol. 13, no. 3, pp. 556–564, Jun. 2006.
- [407] C. Perrier, A. Beroual, and J.-L. Bessede, "Experimental investigations on different insulating liquids and mixtures for power transformers," in *Proc. IEEE Int. Symp. Electrical Insulation*, Sep. 2004, pp. 237–240.
- [408] C. P. McShane, J. Luksich, and K. J. Rapp, "Retrofilling aging transformers with natural ester based dielectric coolant for safety and life extension," in *Proc. IEEE Cement Industry Technical Conf.*, May 2003, pp. 141–147.
- [409] A. Takizawa, Y. Ono, S. Inaka, K. Konno, Y. Nozaki, T. Suzuki, S. Yamazaki, S. Saito, and T. Shirone, "Development of large capacity low-noise gas-insulated transformer," in *Proc. IEEE Power Eng. Soc.*, 1999, vol. 2, pp. 1036–1041.
- [410] S. Trabulus, "Design criteria for SF6 gas insulated power transformers up to 2500 kVA," in *Proc. Annu. Rep. Conf. Electrical Insulation and Dielectric Phenomena*, 1999, vol. 1, pp. 321–326.
- [411] A. R. G. Castro and V. Miranda, "Knowledge discovery in neural networks with application to transformer failure diagnosis," *IEEE Trans. Power Syst.*, vol. 20, no. 2, pp. 717–724, May 2005.
- [412] X. Hao and S. Cai-xin, "Artificial immune network classification algorithm for fault diagnosis of power transformer," *IEEE Trans. Power Del.*, vol. 22, no. 2, pp. 930–935, Apr. 2007.
- [413] J. L. Guardado, J. L. Naredo, P. Moreno, and C. R. Fuerte, "A comparative study of neural network efficiency in power transformers diagnosis using dissolved gas analysis," *IEEE Trans. Power Del.*, vol. 16, no. 4, pp. 643–647, Oct. 2001.
- [414] A. S. Farag, M. Mohandes, and A. Al-Shaikh, "Diagnosing failed distribution transformers using neural networks," *IEEE Trans. Power Del.*, vol. 16, no. 4, pp. 631–636, Oct. 2001.
- [415] Y. C. Huang, "Evolving neural nets for fault diagnosis of power transformers," *IEEE Trans. Power Del.*, vol. 18, no. 3, pp. 843–848, Jul. 2003.
- [416] J. K. Pylvänäinen, K. Nousiainen, and P. Verho, "Studies to utilize loading guides and ANN for oil-immersed distribution transformer condition monitoring," *IEEE Trans. Power Del.*, vol. 22, no. 1, pp. 201–207, Jan. 2007.
- [417] P. S. Georgilakis, J. A. Katsigiannis, K. P. Valavanis, and A. T. Souflaris, "A systematic stochastic petri net based methodology for transformer fault diagnosis and repair actions," *J. Intell. Robot. Syst.*, vol. 45, no. 2, pp. 181–201, 2006.
- [418] W. H. Tang, S. He, Q. H. Wu, and Z. J. Richardson, "Winding deformation identification using a particle swarm optimizer with passive congregation for power transformers," *Int. J. Innovations Energy Syst. Power*, vol. 1, no. 1, pp. 46–52, Nov. 2006.
- [419] Z. J. Richardson, J. Fitch, W. H. Tang, J. Y. Goulermas, and O. H. Wu, "A probabilistic classifier for transformer dissolved gas analysis with a particle swarm optimizer," *IEEE Trans. Power Del.*, vol. 23, no. 2, pp. 751–759, Apr. 2008.
- [420] Y. Hong-Tzer and L. Chiung-Chou, "Adaptive fuzzy diagnosis system for dissolved gas analysis of power transformers," *IEEE Trans. Power Del.*, vol. 14, no. 4, pp. 1342–1350, Oct. 1999.
- [421] T. K. Saha and P. Purkait, "Investigation of an expert system for the condition assessment of transformer insulation based on dielectric response measurements," *IEEE Trans. Power Del.*, vol. 19, no. 3, pp. 1127–1134, Jul. 2004.

- [422] D. Q. Bi, X. A. Zhang, H. H. Yang, G. W. Yu, X. H. Wang, and W. J. Wang, "Correlation analysis of waveforms in nonsaturation zone-based method to identify the magnetizing inrush in transformer," *IEEE Trans. Power Del.*, vol. 22, no. 3, pp. 1380–1385, Jul. 2007.
- [423] C. Bartoletti, M. Desiderio, D. Di Carlo, G. Fazio, F. Muzi, G. Sacerdoti, and F. Salvatori, "Vibro-acoustic techniques to diagnose power transformers," *IEEE Trans. Power Del.*, vol. 19, no. 1, pp. 221–229, Jan. 2004.
- [424] B. García, J. C. Burgos, A. M. Alonso, and J. Sanz, "A moisture-in-oil model for power transformer monitoring—Part I: Theoretical foundation," *IEEE Trans. Power Del.*, vol. 20, no. 2, pt. 2, pp. 1417–1422, Apr. 2005.
- [425] M. Wang, A. J. Vandermaar, and K. D. Srivastava, "Review of condition assessment of power transformers in service," *IEEE Electr. Insul. Mag.*, vol. 18, no. 6, pp. 12–25, Nov. 2002.



Eleftherios I. Amoiralis (M'09) was born in Greece in 1980. He received the Diploma in production and management engineering, the M.Sc. in industrial engineering, and the Ph.D. degree in the field of electric power systems from the Technical University of Crete (TUC), Greece, in 2004, 2005, and 2008, respectively.

Since 2005, he has been a Freelancer with Schneider Electric AE. In 2001 and 2002, he was an Intern at the Public Power Cooperation, Greece. Since 2004, he has a been Research Associate for

many research projects. His current research interests include transformer cost evaluation, energy-efficient transformers, optimal transformer sizing, transformer design optimization, as well as artificial intelligence.

Dr. Amoiralis is a member of the Technical Chamber of Greece.



Marina A. Tsili (M'04) was born in Greece in 1976. She received the Diploma in electrical and computer engineering and the Ph.D. degree from the National Technical University of Athens, Athens, Greece, in 2001 and 2005, respectively.

From 2005 to 2006, she was with the Distribution Division of the Public Power Corporation of Greece in high- and medium-voltage substation studies. In 2007, she joined the Hellenic Transmission System Operator as a Power Systems Engineer. Her research interests include transformer and electric machine

modeling as well as the analysis of generating units by renewable energy sources.

Dr. Tsili is a member of the Technical Chamber of Greece.



Antonios G. Kladas (M'02) was born in Greece in 1959. He received the Diploma in electrical engineering from the Aristotle University of Thessaloniki, Thessaloniki, Greece, in 1982 and the D.E.A. and Ph.D. degrees in electrical engineering from the University of Pierre and Marie Curie (Paris 6), Paris, France, in 1983 and 1987, respectively. He was Associate Assistant at the University of Pierre and Marie Curie from 1984 to 1989. From 1991 to 1996, he joined the Public Power Corporation of Greece, where he was engaged in the System

Studies Department. Since 1996, he has been with the Department of Electrical and Computer Engineering of the National Technical University of Athens (NTUA), where he is now Professor. His research interests include transformer and electric machine modeling and design as well as the analysis of generating units by renewable energy sources and industrial drives.